fundamentals of INVEST

Valuation and Management

seventh edition

MENTS

BRADFORD D. JORDAN | THOMAS W. MILLER JR. | STEVEN D. DOLVIN

Fundamentals of Investments VALUATION AND MANAGEMENT

The McGraw-Hill/Irwin Series in Finance, Insurance, and Real Estate

Stephen A. Ross

Franco Modigliani Professor of Finance and Economics, Sloan School of Management, Massachusetts Institute of Technology, Consulting Editor

Financial Management

Block, Hirt, and Danielsen Foundations of Financial Management Fifteenth Edition

Brealey, Myers, and Allen Principles of Corporate Finance Eleventh Edition

Brealey, Myers, and Allen Principles of Corporate Finance, Concise Second Edition

Brealey, Myers, and Marcus Fundamentals of Corporate Finance Seventh Edition

Brooks FinGame Online 5.0

Bruner Case Studies in Finance: Managing for Corporate Value Creation Seventh Edition

Cornett, Adair, and Nofsinger Finance: Applications and Theory Third Edition

Cornett, Adair, and Nofsinger M: Finance Second Edition

DeMello Cases in Finance Second Edition

Grinblatt (editor) Stephen A. Ross, Mentor: Influence through Generations

Grinblatt and Titman Financial Markets and Corporate Strategy Second Edition

Higgins Analysis for Financial Management Tenth Edition

Kellison Theory of Interest Third Edition

Ross, Westerfield, and Jaffe Corporate Finance Tenth Edition

Ross, Westerfield, Jaffe, and Jordan Corporate Finance: Core Principles and Applications Fourth Edition Ross, Westerfield, and Jordan Essentials of Corporate Finance Eighth Edition Ross, Westerfield, and Jordan Fundamentals of Corporate Finance Tenth Edition Shefrin Behavioral Corporate Finance: Decisions That Create Value First Edition White Financial Analysis with an Electronic Calculator Sixth Edition

Investments

Bodie, Kane, and Marcus Essentials of Investments Ninth Edition Bodie Kane and Marcus Investments Tenth Edition Hirt and Block Fundamentals of Investment Management Tenth Edition Iordan and Miller Fundamentals of Investments: Valuation and Management Seventh Edition Stewart, Piros, and Heisler Running Money: Professional Portfolio Management First Edition Sundaram and Das Derivatives: Principles and Practice First Edition

Financial Institutions and Markets

Rose and Hudgins Bank Management and Financial Services Ninth Edition Rose and Marquis Financial Institutions and Markets Eleventh Edition Saunders and Cornett Financial Institutions Management: A Risk Management Approach Eighth Edition Saunders and Cornett Financial Markets and Institutions Fifth Edition

International Finance

Eun and Resnick International Financial Management Seventh Edition

Real Estate

Brueggeman and Fisher Real Estate Finance and Investments Fourteenth Edition

Ling and Archer Real Estate Principles: A Value Approach Fourth Edition

Financial Planning and Insurance

Allen, Melone, Rosenbloom, and Mahoney Retirement Plans: 401(k)s, IRAs, and Other Deferred Compensation Approaches Eleventh Edition

Altfest Personal Financial Planning First Edition

Harrington and Niehaus Risk Management and Insurance Second Edition

Kapoor, Dlabay, and Hughes Focus on Personal Finance: An Active Approach to Help You Develop Successful Financial Skills Fourth Edition

Kapoor, Dlabay, and Hughes Personal Finance Eleventh Edition

Walker and Walker Personal Finance: Building Your Future First Edition Seventh Edition

Fundamentals of Investments VALUATION AND MANAGEMENT

Bradford D. Jordan

University of Kentucky

Thomas W. Miller Jr. Mississippi State University

Steven D. Dolvin, CFA

Butler University





FUNDAMENTALS OF INVESTMENTS: VALUATION AND MANAGEMENT, SEVENTH EDITION

Published by McGraw-Hill Education, 2 Penn Plaza, New York, NY 10121. Copyright © 2015 by McGraw-Hill Education. All rights reserved. Printed in the United States of America. Previous editions © 2012, 2009, and 2008. No part of this publication may be reproduced or distributed in any form or by any means, or stored in a database or retrieval system, without the prior written consent of McGraw-Hill Education, including, but not limited to, in any network or other electronic storage or transmission, or broadcast for distance learning.

Some ancillaries, including electronic and print components, may not be available to customers outside the United States.

This book is printed on acid-free paper.

 $1\ 2\ 3\ 4\ 5\ 6\ 7\ 8\ 9\ 0$ DOW/DOW $1\ 0\ 9\ 8\ 7\ 6\ 5\ 4$

ISBN 978-0-07-786163-6 MHID 0-07-786163-9

Senior Vice President, Products & Markets: Kurt L. Strand Vice President, Content Production & Technology Services: Kimberly Meriwether David Managing Director: Douglas Reiner Executive Brand Manager: Chuck Synovec Executive Director of Development: Ann Torbert Development Editor II: Jennifer Lohn Upton Director of Digital Content: Doug Ruby Digital Development Editor: Kevin Shanahan Executive Marketing Manager: Melissa S. Caughlin Director, Content Production: Terri Schiesl Content Project Manager: Brian Nacik Senior Buyer: Debra R. Sylvester Design: Matt Diamond Cover Image: Veer Images Typeface: 10/12 Times Roman Compositor: MPS Limited Printer: R. R. Donnelley

All credits appearing on page or at the end of the book are considered to be an extension of the copyright page.

Library of Congress Cataloging-in-Publication Data

Jordan, Bradford D.

Fundamentals of investments : valuation and management / Bradford D. Jordan, University of
Kentucky, Thomas W. Miller Jr., Mississippi State, Steven D. Dolvin, CFA, Butler University.—Seventh edition. pages cm. — (The McGraw-Hill/Irwin series in finance, insurance, and real estate)
Includes index.
ISBN 978-0-07-786163-6 (alk. paper)—ISBN 0-07-786163-9 (alk. paper)
1. Investments. I. Miller, Thomas W. II. Dolvin, Steven D. III. Title.
HG4521.C66 2015
332.6—dc23

2013039057

The Internet addresses listed in the text were accurate at the time of publication. The inclusion of a website does not indicate an endorsement by the authors or McGraw-Hill Education, and McGraw-Hill Education does not guarantee the accuracy of the information presented at these sites.

To my late father, S. Kelly Jordan Sr., a great stock picker.

BDJ

To my parents, Tom and Kathy Miller, my wife Carolyn, and #21 —Thomas W. Miller III.

TWM Jr.

To my wife, Kourtney, and the "three L's"—my greatest investment in this life.

SDD

About the Authors

Bradford D. Jordan

Gatton College of Business and Economics, University of Kentucky

Bradford D. Jordan is Professor of Finance and holder of the Richard W. and Janis H. Furst Endowed Chair in Finance at the University of Kentucky. He has a long-standing interest in both applied and theoretical issues in investments, and he has extensive experience teaching all levels of investments. Professor Jordan has published numerous research articles on issues such as valuation of fixed-income securities, tax effects in investments analysis, the behavior of security prices, IPO valuation, and pricing of exotic options. He is co-author of *Fundamentals of Corporate Finance* and *Essentials of Corporate Finance*, two of the most widely used finance textbooks in the world.

Thomas W. Miller Jr.

College of Business, Mississippi State University

Tom Miller is Professor of Finance and holder of the Jack R. Lee Chair in Financial and Consumer Finance at Mississippi State University. Professor Miller has a long-standing interest in derivative securities and investments and has published numerous articles on various topics in these areas. Professor Miller has been honored with many research and teaching awards. Professor Miller is a co-author (with David Dubofsky) of *Derivatives: Valuation and Risk Management* (Oxford University Press). Professor Miller's interests include golf, skiing, American saddlebred horses, and playing tenor saxophone.

Steven D. Dolvin, CFA

College of Business, Butler University

Steven D. Dolvin, CFA, is an Associate Professor of Finance at Butler University. He teaches primarily in the area of investments, but he also oversees student-run portfolios in both public and private equity. He has received multiple teaching awards and has also published numerous articles in both academic and practitioner outlets. His principal areas of interest are IPOs, venture capital, financial education, retirement investing, and behavioral finance. His prior experience includes work in both corporate finance and investments, and he currently does investment consulting for both individuals and businesses. Professor Dolvin is also a CFA charterholder and is actively involved in his local CFA society.

Preface

So why *did* we write this book?

As we toiled away, we asked ourselves this question many times, and the answer was always the same: *Our students made us*.

Traditionally, investments textbooks tend to fall into one of two camps. The first type has a greater focus on portfolio management and covers a significant amount of portfolio theory. The second type is more concerned with security analysis and generally contains fairly detailed coverage of fundamental analysis as a tool for equity valuation. Today, most texts try to cover all the bases by including some chapters drawn from one camp and some from another.

The result of trying to cover everything is either a very long book or one that forces the instructor to bounce back and forth between chapters. This frequently leads to a noticeable lack of consistency in treatment. Different chapters have completely different approaches: Some are computational, some are theoretical, and some are descriptive. Some do macroeconomic forecasting, some do mean-variance portfolio theory and beta estimation, and some do financial statements analysis. Options and futures are often essentially tacked on the back to round out this disconnected assortment.

The goal of these books is different from the goal of our students. Our students told us they come into an investments course wanting to learn how to make investment decisions. As time went by, we found ourselves supplying more and more supplemental materials to the texts we were using and constantly varying chapter sequences while chasing this elusive goal. We finally came to realize that the financial world had changed tremendously, and investments textbooks had fallen far behind in content and relevance.

What we really wanted, and what our students really needed, was a book that would do several key things:

- Focus on the students as investment managers by giving them information they can
 act on instead of concentrating on theories and research without the proper context.
- Offer strong, consistent pedagogy, including a balanced, unified treatment of the main types of financial investments as mirrored in the investment world.
- Organize topics in a way that would make them easy to apply—whether to a portfolio simulation or to real life—and support these topics with hands-on activities.

We made these three goals the guiding principles in writing this book. The next several sections explain our approach to each and why we think they are so important.

Who Is This Book For?

This book is aimed at introductory investments classes with students who have relatively little familiarity with investments. A typical student may have taken a principles of finance class and had some exposure to stocks and bonds, but not much beyond the basics. The introductory investments class is often a required course for finance majors, but students from other areas often take it as an elective. One fact of which we are acutely aware is that this may be the only investments class many students will ever take.

We intentionally wrote this book in a relaxed, informal style that engages the student and treats him or her as an active participant rather than a passive information absorber. We think the world of investments is exciting and fascinating, and we hope to share our considerable enthusiasm for investing with the student. We appeal to intuition and basic principles whenever possible because we have found that this approach effectively promotes understanding. We also make extensive use of examples throughout, drawing on material from the world around us and using familiar companies wherever appropriate.

By design, the text is not encyclopedic. As the table of contents indicates, we have a total of 20 chapters. Chapter length is about 30 to 40 pages, so the text is aimed at a single-term course; most of the book can be covered in a typical quarter or semester.

Aiming the book at a one-semester course necessarily means some picking and choosing, with regard to both topics and depth of coverage. Throughout, we strike a balance by introducing and covering the essentials while leaving some of the details to follow-up courses in security analysis, portfolio management, and options and futures.

How Does the Seventh Edition of This Book Expand upon the Goals Described Above?

Based on user feedback, we have made numerous improvements and refinements in the seventh edition of *Fundamentals of Investments: Valuation and Management*. We updated an appendix containing useful formulas. We updated every chapter to reflect current market practices and conditions, and we significantly expanded and improved the end-of-chapter material. Also, our chapters devoted to market efficiency and to behavioral finance continue to rate highly among readers.

To give some examples of our additional new content:

- Chapter 1 contains updates on historical returns for small-company stocks, large-company stocks, long-term government bonds, Treasury bills, as well as U.S. inflation rates.
- Chapter 2 contains new material on AAII asset allocation models.
- · Chapter 3 incorporates the new ticker symbols for exchange-traded options.
- Chapter 4 contains new material on the key difference between two popular S&P 500 ETFs.
- Chapter 5 contains new material on the Flash Crash of 2010 as well as updated material on circuit breakers.
- Chapter 6 contains a new section on enterprise value ratios. It also contains a detailed new example showing how to value Procter & Gamble Company using the models presented in the chapter.
- Chapter 8 contains new material on why investors find it difficult to sell losers. Students have an opportunity to take an online quiz about overconfidence.
- Chapter 11 contains new material on the fallacy of time diversification.
- Chapter 13 contains new material on the Sortino ratio.
- Chapter 15 contains new material on weekly options. The chapter also has updated material on credit default swaps (CDSs).
- Chapter 17 contains an updated valuation for Starbucks Corporation.
- Chapter 18 combines material on corporate, U.S. federal government, and municipal bonds previously contained in two separate chapters.
- Chapter 19 is a new chapter on global economic activity and industry analysis. This
 new chapter contains material relevant to investors striving to identify how best to
 allocate their portfolio weights.

In addition, we have updated learning objectives for each chapter. We have reworked our chapter summaries to reflect the chapter's learning objectives.

For the seventh edition, we significantly expanded and improved the end-of-chapter material. We added new problems throughout, and we have significantly increased the CFATM content. We updated the questions that test understanding of concepts with no calculations involved. Additionally, our *What's on the Web?* questions give students assignments

to perform based on information they retrieve from various Web sites. Finally, in selected chapters, we have included spreadsheet assignments, which ask students to create certain types of spreadsheets to solve problems.

We continue to emphasize the use of the Web in investments analysis, and we integrate Web-based content in several ways. First, wherever appropriate, we provide a commented link in the margin. These links send readers to selected, particularly relevant Web sites. Second, our *Work the Web* feature, expanded and completely updated for this edition, appears in most chapters. These boxed readings use screen shots to show students how to access, use, and interpret various types of key financial and market data. Finally, as previously noted, new end-of-chapter problems rely on data retrieved from the Web.

We continue to provide *Spreadsheet Analysis* exhibits, which we have enhanced for this edition. These exhibits illustrate directly how to use spreadsheets to do certain types of important problems, including such computationally intensive tasks as calculating Macaulay duration, finding Black-Scholes option prices, and determining optimal portfolios based on Sharpe ratios. We also continue to provide, where relevant, readings from *The Wall Street Journal*, which have been thoroughly updated for this edition.

CFA[™] Mapping

Consider this description provided by the CFA Institute: "First awarded in 1963, the Chartered Financial Analyst (CFA) charter has become known as the gold standard of professional credentials within the global investment community. Investors recognize the CFA designation as the definitive standard for measuring competence and integrity in the fields of portfolio management and investment analysis." The importance and growing significance of the CFA charter are compelling reasons to integrate CFA curriculum material into our seventh edition.

Among the requirements to earn the CFA charter, candidates must pass three sequential levels of comprehensive exams. Each exam asks questions on a wide array of subject areas concerning the investment process. To help candidates study for the exams, the exams at each level are divided into so-called study sessions. Each of these study sessions has a core set of readings designed to help prepare the candidate for the exams. We carefully examined the content of each reading (updated for the 2012 exams), as well as the stated learning outcomes, to determine which areas we covered in the sixth edition. Importantly, we also considered which areas might be added to the seventh edition.

As a result of this thorough process, in our seventh edition we expanded coverage on seven readings and added completely new coverage of three readings. In total, our textbook contains material that touches over 75 percent of the readings from Level 1 of the CFA exam. Topics that we do not address from Level 1, such as basic statistics, accounting, and economics, are likely addressed in prerequisite courses taken before the investments course. In addition, we present some higher-level material: We touch on about 35 percent of the readings from the Level 2 and 3 exams.

Of course, we make no claim that our textbook is a substitute for the CFA exam readings. Nonetheless, we believe that our seventh edition provides a terrific framework and introduction for students looking to pursue a career in investments—particularly for those interested in eventually holding the CFA charter. To provide a sense of studying for the CFA, the seventh edition continues to include an end-of-chapter case review. *Schweser*, a leading purveyor of CFA exam preparation packages, graciously provided extensive material from which we chose these case reviews.

We provide a mapping between the textbook and the CFA curriculum as follows: Each chapter opens with a CFA Exam box citing references to specific readings from the CFA curriculum that are covered within the chapter. The topic is identified and we indicate which level and study session the reading comes from. We label these topics CFA1, CFA2, CFA3, and so on, for easy reference. End-of-chapter problems in the book and in *Connect* are also labeled with these tags. Over 95 percent of our end-of-chapter material is related to the CFA exam. We believe that this integration adds tremendous value to the seventh edition.

Assurance-of-Learning Ready

Many educational institutions today are focused on the notion of assurance of learning, an important element of some accreditation standards. This edition is designed specifically to support your assurance-of-learning initiatives with a simple, yet powerful, solution. Listed below are the learning objectives for each chapter.

Each test bank question for this book maps to a specific chapter learning objective listed in the text. You can use the test bank software to easily query for learning outcomes and objectives that directly relate to the learning objectives for your course. You can then use the reporting features of the software to aggregate student results in similar fashion, making the collection and presentation of assurance-of-learning data simple and easy.

Chapter Learning Objectives

Chapter 1: A Brief History of Risk and Return

To become a wise investor (maybe even one with too much money), you need to know:

- 1. How to calculate the return on an investment using different methods.
- 2. The historical returns on various important types of investments.
- 3. The historical risks on various important types of investments.
- 4. The relationship between risk and return.

Chapter 2: The Investment Process

Don't sell yourself short. Instead, learn about these key investment subjects:

- **1.** The importance of an investment policy statement.
- 2. The various types of securities brokers and brokerage accounts.
- 3. How to calculate initial and maintenance margin.
- **4.** The workings of short sales.

Chapter 3: Overview of Security Types

Price quotes for all types of investments are easy to find, but what do they mean? Learn the answers for:

- **1.** Various types of interest-bearing assets.
- 2. Equity securities.
- 3. Futures contracts.
- 4. Option contracts.

Chapter 4: Mutual Funds and Other Investment Companies

You're probably going to be a mutual fund investor very soon, so you should definitely know the following:

- **1.** The different types of mutual funds.
- 2. How mutual funds operate.
- 3. How to find information about how mutual funds have performed.
- 4. The workings of exchange-traded funds (ETFs) and hedge funds.

Chapter 5: The Stock Market

Take stock in yourself. Make sure you have a good understanding of:

- **1.** The differences between private and public equity and between primary and secondary stock markets.
- 2. The workings of the New York Stock Exchange.
- 3. How NASDAQ operates.
- 4. How to calculate index returns.

Chapter 6: Common Stock Valuation

Separate yourself from the commoners by having a good understanding of these security valuation methods:

- 1. The basic dividend discount model.
- **2.** The two-stage dividend growth model.
- 3. The residual income and free cash flow models.
- 4. Price ratio analysis.

Chapter 7: Stock Price Behavior and Market Efficiency

You should strive to have your investment knowledge fully reflect:

- 1. The foundations of market efficiency.
- 2. The implications of the forms of market efficiency.
- 3. Market efficiency and the performance of professional money managers.
- 4. What stock market anomalies, bubbles, and crashes mean for market efficiency.

Chapter 8: Behavioral Finance and the Psychology of Investing

Psych yourself up and get a good understanding of:

- 1. Prospect theory.
- 2. The implications of investor overconfidence and misperceptions of randomness.
- 3. Sentiment-based risk and limits to arbitrage.
- 4. The wide array of technical analysis methods used by investors.

Chapter 9: Interest Rates

It will be worth your time to increase your rate of interest in these topics:

- 1. Money market prices and rates.
- 2. Rates and yields on fixed-income securities.
- **3.** Treasury STRIPS and the term structure of interest rates.
- 4. Nominal versus real interest rates.

Chapter 10: Bond Prices and Yields

Bonds can be an important part of portfolios. You will learn:

- 1. How to calculate bond prices and yields.
- 2. The importance of yield to maturity.
- 3. Interest rate risk and Malkiel's theorems.
- 4. How to measure the impact of interest rate changes on bond prices.

Chapter 11: Diversification and Risky Asset Allocation

To get the most out of this chapter, diversify your study time across:

- 1. How to calculate expected returns and variances for a security.
- 2. How to calculate expected returns and variances for a portfolio.
- 3. The importance of portfolio diversification.
- 4. The efficient frontier and the importance of asset allocation.

Chapter 12: Return, Risk, and the Security Market Line

Studying some topics will yield an expected reward. For example, make sure you know:

- 1. The difference between expected and unexpected returns.
- 2. The difference between systematic risk and unsystematic risk.
- **3.** The security market line and the capital asset pricing model.
- **4.** The importance of beta.

Chapter 13: Performance Evaluation and Risk Management

To get a high evaluation of your investments' performance, make sure you know:

- 1. How to calculate the best-known portfolio evaluation measures.
- 2. The strengths and weaknesses of these portfolio evaluation measures.
- 3. How to calculate a Sharpe-optimal portfolio.
- 4. How to calculate and interpret Value-at-Risk.

Chapter 14: Futures Contracts

You will derive many future benefits if you have a good understanding of:

- **1.** The basics of futures markets and how to obtain price quotes for futures contracts.
- 2. The risks involved in futures market speculation.
- **3.** How cash prices and futures prices are linked.
- 4. How futures contracts can be used to transfer price risk.

Chapter 15: Stock Options

Give yourself some in-the-money academic and professional options by understanding:

- 1. The basics of option contracts and how to obtain price quotes.
- **2.** The difference between option payoffs and option profits.
- 3. The workings of some basic option trading strategies.
- **4.** The logic behind the put-call parity condition.

Chapter 16: Option Valuation

Make sure the price is right by making sure that you have a good understanding of:

- 1. How to price options using the one-period and two-period binomial models.
- 2. How to price options using the Black-Scholes model.
- **3.** How to hedge a stock portfolio using options.
- 4. The workings of employee stock options.

Chapter 17: Projecting Cash Flow and Earnings

Help yourself grow as a stock analyst by knowing:

- 1. How to obtain financial information about companies.
- **2.** How to read basic financial statements.
- **3.** How to use performance and price ratios.
- **4.** How to use the percentage of sales method in financial forecasting.

Chapter 18: Corporate and Government Bonds

Conform to your fixed-income knowledge covenants by learning:

- **1.** The basic types of corporate bonds.
- 2. How callable and convertible bonds function.
- **3.** The different types of government bonds.
- 4. The basics of bond ratings.

Chapter 19: Global Economic Activity and Industry Analysis

If you want the supply of your investment services to be in high demand, you should:

- 1. Understand the process of top-down analysis.
- 2. Be able to measure the level of economic activity globally and domestically.
- 3. Understand the relation of monetary and fiscal policies to economic activity.
- 4. Be able to identify industry sensitivity to business cycles.

Chapter 20 (Web site only): Mortgage-Backed Securities

Before you mortgage your future, you should know:

- 1. The workings of a fixed-rate mortgage.
- 2. Government's role in the secondary market for home mortgages.
- 3. The impact of mortgage prepayments.
- 4. How collateralized mortgage obligations are created and divided.

How Is This Book Relevant to the Student?

Fundamental changes in the investments universe drive our attention to relevance. The first major change is that individuals are being asked to make investment decisions for their own portfolios more often than ever before. There is, thankfully, a growing recognition that traditional "savings account" approaches to investing are decidedly inferior. At the same time, the use of employer-sponsored "investment accounts" has expanded enormously. The second major change is that the investments universe has exploded with an ever-increasing number of investment vehicles available to individual investors. As a result, investors must choose from an array of products, many of which are very complex, and they must strive to choose wisely.

Beyond this, students are more interested in subjects that affect them directly (as are we all). By taking the point of view of the student as an investor, we are better able to illustrate and emphasize the relevance and importance of the material.

Our approach is evident in the table of contents. Our first chapter is motivational; we have found that this material effectively "hooks" students and even motivates a semester-long discourse on risk and return. Our second chapter answers the student's next natural question: "How do I get started investing and how do I buy and sell securities?" The third chapter surveys the different types of investments available. After only three chapters, very early in the term, students have learned something about the risks and rewards from investing, how to get started investing, and what investment choices are available.

We close the first part of the text with a detailed examination of mutual funds. Without a doubt, mutual funds have become the most popular investment vehicles for individual investors. There are now more mutual funds than there are stocks on the NYSE! Given the size and enormous growth in the mutual fund industry, this material is important for investors. Even so, investments texts typically cover mutual funds in a cursory way, often banishing the material to a back chapter under the obscure (and obsolete) heading of "investment companies." Our early placement lets students quickly explore a topic they have heard a lot about and are typically interested in learning more about.

How Does This Book Allow Students to Apply the Investments Knowledge They Learn?

After studying this text, students will have the basic knowledge needed to move forward and actually act on what they have learned. We have developed two features to encourage students in making decisions as an investment manager. Learning to make good investment decisions comes with experience, while experience (regrettably) comes from making bad investment decisions. As much as possible, we press our students to get those bad decisions out of their systems before they start managing real money!

Not surprisingly, most students don't know how to get started in buying and selling securities. We have learned that providing some structure, especially with a portfolio simulation, greatly enhances the experience. Therefore, we have a series of *Getting Down to Business* boxes. These boxes (at the end of each chapter) usually describe actual trades for students to explore. The intention is to show students how to gain real experience with the principles and instruments covered in the chapter. The second feature is a series of *Stock-Trak* exercises that take students through specific trading situations using *Stock-Trak Portfolio Simulations*, which can be found at the book's Web site, www.mhhe.com/jmd7e.

Because we feel that portfolio simulations are so valuable, we have taken steps to assist instructors who, like us, plan to integrate portfolio simulations into their courses. Beyond the features mentioned above, we have organized the text so that the essential material needed before participating in a simulation is covered at the front of the book. Most notably, with every book, we have included a *free* subscription to *Stock-Trak Portfolio Simulations. Stock-Trak* is the leading provider of investment simulation services to the academic community; providing *Stock-Trak* free represents a significant cost savings to students. To our knowledge, ours is the first (and only) investments text to directly offer a full-featured online brokerage account simulation with the book at no incremental cost.

How Does This Book Maintain a Consistent, Unified Treatment?

In most investments texts, depth of treatment and presentation vary dramatically from instrument to instrument, which leaves the student without an overall framework for understanding the many types of investments. We stress early on that there are essentially only four basic types of financial investments—stocks, bonds, options, and futures. In Parts 2 through 6, our simple goal is to take a closer look at each of these instruments. We take a unified approach to each by answering these basic questions:

- 1. What are the essential features of the instrument?
- 2. What are the possible rewards?
- 3. What are the risks?
- 4. What are the basic determinants of investment value?
- 5. For whom is the investment appropriate and under what circumstances?
- 6. How is the instrument bought and sold, and how does the market for the instrument operate?

By covering investment instruments in this way, we teach the students what questions to ask when looking at any potential investment.

Unlike other introductory investments texts, we devote several chapters beyond the basics to the different types of fixed-income investments. Students are often surprised to learn that the fixed-income markets are so much bigger than the equity markets and that money management opportunities are much more common in the fixed-income arena. Possibly the best way to see this is to look at recent CFA exams and materials and note the extensive coverage of fixed-income topics. We have placed these chapters toward the back of the text because we recognize not everyone will want to cover all this material. We have also separated the subject into several shorter chapters to make it more digestible for students and to allow instructors more control over what is covered.

Acknowledgments

We have received extensive feedback from reviewers at each step along the way, and we are very grateful to the following dedicated scholars and teachers for their time and expertise:

Aaron Phillips, California State University–Bakersfield
Adam Schwartz, Washington and Lee University
Alan Wong, Indiana University Southeast
Allan O'Bryan, Rochester Community & Technical College
Allan Zebedee, San Diego State University
Ann Hackert, Idaho State University
Benito Sanchez, Kean University
Bruce Grace, Morehead State University
Carl R. Chen, University of Dayton
Carla Rich, Pensacola Junior College
Caroline Fulmer, University of Alabama
Charles Appeadu, University of North Florida

Christos Giannikos, Bernard M. Baruch College Crystal Ayers, College of Southern Idaho David Dubofsky, University of Louisville David Hunter, University of Hawaii-Manoa David Louton, Bryant College David Loy, Illinois State University David Peterson, Florida State University David Stewart. Winston-Salem State University Deborah Murphy, University of Tennessee-Knoxville Dina Layish, Binghamton University Donald Wort, California State University-East Bay Donald Lennard, Park University Dwight Giles, Jefferson State Community College Edward Miller, University of New Orleans Felix Ayadi, Fayetteville State University Gary Engle, University of Wisconsin-Milwaukee Gay B. Hatfield, University of Mississippi George Jouganatos, California State University-Sacramento Gioia Bales, Hofstra University Haigang Zhou, Cleveland State University Howard Van Auken, Iowa State University Howard W. Bohnen, St. Cloud State University Imad Elhaj, University of Louisville It-Keong Chew, University of Kentucky James Forjan, York College of Pennsylvania Jeff Brookman, Idaho State University Jeff Edwards, Portland Community College Jeff Manzi, Ohio University Jennifer Morton, Ivy Technical Community College of Indiana Ji Chen, University of Colorado Jim Tipton, Baylor University Joan Anderssen, Arapahoe Community College Joe Brocato, Tarleton State University Joe Walker, University of Alabama–Birmingham John Bockino, Suffolk County Community College John Clinebell, University of Northern Colorado John Finnigan, Marist College John Ledgerwood, Bethune-Cookman College John Paul Broussard, Rutgers, The State University of New Jersey John Romps, St. Anselm College John Stocker, University of Delaware John Wingender, Creighton University Johnny Chan, University of Dayton Jorge Omar R. Brusa, University of Arkansas Karen Bonding, University of Virginia Keith Fevurly, Metropolitan State College of Denver Kerri McMillan, Clemson University Ladd Kochman, Kennesaw State University

Lalatendu Misra, University of Texas at San Antonio Lawrence Blose, Grand Valley State University Linda Martin, Arizona State University Lisa Schwartz, Wingate University M. J. Murray, Winona State University Majid R. Muhtaseb, California State Polytechnic University Marc LeFebvre, Creighton University Marie Kratochvil, Nassau Community College Margo Kraft, Heidelberg College Matthew Fung, Saint Peter's College Michael C. Ehrhardt, University of Tennessee-Knoxville Michael Gordinier, Washington University Michael Milligan, California State University-Northridge Michael Nugent, SUNY-Stony Brook Mukesh Chaudhry, Indiana University of Pennsylvania Naresh Bansal, Saint Louis University Nolan Lickey, Utah Valley State College Nozar Hashemzadeh, Radford University Patricia Clarke, Simmons College Paul Bolster, Northeastern University Percy S. Poon, University of Nevada, Las Vegas Ping Hsao, San Francisco State University Praveen K. Das, University of Louisiana-Lafayette Rahul Verma, University of Houston Randall Wade, Rogue Community College Richard Followill, University of Northern Iowa Richard Lee Kitchen, Tallahassee Community College Richard Proctor, Siena College Richard W. Taylor, Arkansas State University Robert Friederichs, Alexandria Technical College Robert Kao, Park University Robert Kozub, University of Wisconsin-Milwaukee Robert L. Losey, University of Louisville Ronald Christner, Lovola University-New Orleans Samira Hussein, Johnson County Community College Sammie Root, Texas State University-San Marcos Samuel H. Penkar, University of Houston Scott Barnhart, Clemson University Scott Beyer, University of Wisconsin-Oshkosh Scott Gruner, Trine University Stephen Chambers, Johnson County Community College Steven Lifland, High Point University Stuart Michelson, University of Central Florida Thomas M. Krueger, University of Wisconsin–La Crosse Thomas Willey, Grand Valley State University Tim Samolis, Pittsburgh Technical Institute Vernon Stauble, San Bernardino Valley College Ward Hooker, Orangeburg-Calhoun Technical College

William Compton, University of North Carolina–Wilmington
William Elliott, Oklahoma State University
William Lepley, University of Wisconsin–Green Bay
Yvette Harman, Miami University of Ohio
Zekeriah Eser, Eastern Kentucky University

We thank Lynn Kugele, for developing the Test Bank and for creating the Student Narrated PowerPoint slides. We thank R. Douglas Van Eaton, CFA, for providing access to *Schweser's* preparation material for the CFA exam. We would especially like to acknowledge the careful reading and helpful suggestions made by professors John Walker and Frederick Schadler.

Special thanks to Carolyn Moore Miller and Kameron Killian for their efforts. Steve Hailey did outstanding work on this text. To him fell the unenviable task of technical proofreading and, in particular, carefully checking each calculation throughout the supplements.

We are deeply grateful to the select group of professionals who served as our development team on this edition: Chuck Synovec, Executive Brand Manager; Jennifer Upton, Development Editor; Melissa Caughlin, Executive Marketing Manager; Matt Diamond, Designer; Debra Sylvester, Production Supervisor; and Brian Nacik, Content Project Manager.

Bradford D. Jordan Thomas W. Miller Jr. Steven D. Dolvin, CFA

Coverage

This book was designed and developed explicitly for a first course in investments taken either by finance majors or non-finance majors. In terms of background or prerequisites, the book is nearly self-contained, but some familiarity with basic algebra and accounting is assumed. The organization of the text has been designed to give instructors the flexibility they need to teach a quarter-long or semester-long course.

To present an idea of the breadth of coverage in the seventh edition of *Fundamentals of Investments*, the following grid is presented chapter by chapter. This grid contains some of the most significant new features and a few selected chapter highlights. Of course, for each chapter, features like opening vignettes, Work the Web, Spreadsheet Analysis, Getting Down to Business, Investment Updates, tables, figures, examples, and end-of-chapter material has been thoroughly reviewed and updated.

Chapters	Selected Topics of Interest	Learning Outcome/Comment	
PART ONE Introduction			
Chapter 1			
A Brief History of Risk and Return	Dollar returns and percentage returns.	Average returns differ by asset class.	
	Return variability and calculating variance and standard deviation. <i>New material:</i> <i>The best and worst days for the DJIA.</i>	Return variability also differs by asset class.	
	Arithmetic versus geometric returns.	Geometric average tells you what you actually earned per year, compounded annually. Arithmetic returns tell you what you earned in a typical year. Dollar-weighted average returns adjust for investment inflows and outflows.	
	The risk-return trade-off. Updated material: World stock market capitalization.	Historically, higher returns are associated with higher risk. Estimates of future equity risk premiums involve assumptions about the risk environment and investor risk aversion.	
Chapter 2			
The Investment Process	The investment policy statement (IPS).	By knowing their objectives and constraints, investors can capture risk and safety trade-offs in an investment policy statement (IPS).	
	Investor objectives, constraints, and strategies.	Presentation of issues like risk and return, resource constraints, market timing, and asset allocation.	
	Investment professionals and types of brokerage accounts.	Discussion of the different types of financial advisors and brokerage accounts available to an individual investor.	
	Retirement accounts.	Readers will know the workings of company- sponsored plans, such as a 401(k), traditional individual retirement accounts (IRAs), and Roth IRAs.	
	Short sales.	Description of the process of short selling stock and short-selling constraints imposed by regulations and market conditions.	
	Forming an investment portfolio. New material: AAII asset allocation models.	An investment portfolio must account for an investor's risk tolerance, objectives, constraints, and strategies.	

Chapters	Selected Topics of Interest	Learning Outcome/Comment
Chapter 3		
Overview of Security Types	Classifying securities.	Interest-bearing, equity, and derivative securities.
	NASD's new TRACE system and transparency in the corporate bond market.	Up-to-date discussion of new developments in fixed income with respect to price, volume, and transactions reporting.
	Equity securities.	Obtaining price quotes for equity securities.
	Derivative securities: Obtaining futures contract and option contract price quotes using the Internet.	Defining the types of derivative securities, interpreting their price quotes, and calculating gains and losses from these securities.
Chapter 4		
Mutual Funds and Other Investment Companies	Advantages and drawbacks of investing in mutual funds.	Advantages include diversification, professional management, and minimum initial investment. Drawbacks include risk, costs, and taxes.
	Investment companies and types of funds.	Covers concepts like open-end versus closed-end funds and net asset value.
	Mutual fund organization, creation, costs, and fees.	Presents types of expenses and fees like front- end loads, 12b-1 fees, management fees, and turnover.
	Short-term funds, long-term funds, and fund performance.	Discussion of money market mutual funds versus the variety of available stock and bond funds and how to find their performance.
	Special funds like closed-end funds, exchange-traded funds (<i>expanded</i> <i>material on S&P 500 ETFs)</i> , and hedge funds.	The closed-end fund discount mystery and discussion of exchange-traded funds (ETFs), exchange-traded notes (ETNs), hedge fund investment styles, and the perils of leveraged ETFs.
PART TWO Stock Marke	ts	
Chapter 5		
The Stock Market	The primary stock market.	The workings of an initial public offering (IPO), a seasoned equity offering (SEO), the role of investment bankers, and the role of the Securities and Exchange Commission (SEC).
	The secondary stock market. <i>New material:</i> <i>The Flash Crash of 2010. Updated material:</i> <i>Circuit breakers.</i>	The role of dealers and brokers, the workings of the New York Stock Exchange (NYSE), and NASDAQ market operations.
	Stock indexes, including the Dow Jones Industrial Average (DJIA) and the Standard and Poor's 500 Index (S&P 500).	The components of the DJIA and their dividend yields. The difference between price-weighted indexes and value-weighted indexes.
Chapter 6		
Common Stock Valuation	The basic dividend discount model (DDM) and several of its variants, like the two- stage dividend growth model.	Valuation using constant growth rates and nonconstant growth rates.
	The residual income model and the free cash flow model.	Valuation of non-dividend-paying stocks. Valuation of stocks with negative earnings.
	Price ratio analysis.	Valuation using price-earnings, price-cash flow, and price-sales.
	New section: enterprise value ratios.	Valuation of a firm using a ratio containing both debt and equity.
	New material: Valuing Procter & Gamble, a detailed example.	Using Value Line information to value a stock using methods presented earlier in the chapter.

Chapters	Selected Topics of Interest	Learning Outcome/Comment
Chapter 7		
Stock Price Behavior and Market Efficiency	Forms of market efficiency.	The effects of information on stock prices with respect to market efficiency.
	Event studies using actual events surrounding Advanced Medical Optics.	Explains how new information gets into stock prices and how researchers measure it.
	Informed traders, insider trading, and illegal insider trading.	Example: Martha Stewart and ImClone.
	Updated material: Market efficiency and the performance of professional money managers.	Discusses the performance of professional money managers versus static benchmarks.
	Updated material: Anomalies.	Presentation of the day-of-the-week effect, the amazing January effect, the turn-of-the-year effect, and the turn-of-the-month effect.
	Bubbles and crashes. <i>New material:</i> Individual stock circuit breakers.	Shows the extent of famous events like the Crash of 1929, the Crash of October 1987, the Asian market crash, the "dot-com" bubble, and the Crash of 2008.
Chapter 8		
Behavioral Finance and the Psychology of Investing	Introduction to behavioral finance.	The influence of reasoning errors on investor decisions.
	Prospect theory.	How investors tend to behave differently when faced with prospective gains and losses.
	Overconfidence, misperceiving randomness, and overreating to chance events.	Examines the consequences of these serious errors in judgment.
	More on behavioral finance. New material: Letting go of losers.	Heuristics, herding, and overcoming bias.
	Sentiment-based risk and limits to arbitrage.	3Com/Palm mispricing, the Royal Dutch/Shell price ratio.
	Technical analysis.	Advance/decline line indicators, market diary, relative strength charts, and technical analysis data for Microsoft Corp.
PART THREE Interest Rat	es and Bond Valuation	
Chapter 9		
Interest Rates	Interest rate history and a quick review of the time value of money.	A graphical presentation of the long-term history of interest rates.
	Money market rates and their prices.	Important money market concepts including pricing U.S. Treasury bills, bank discount yields versus bond equivalent yields, annual percentage rates, and effective annual returns.
	Rates and yields on fixed-income securities.	The Treasury yield curve, the term structure of interest rates, Treasury STRIPS, and inflation-indexed Treasury securities (TIPS).
	Nominal versus real interest rates.	The Fisher hypothesis.
	Determinants of nominal interest rates.	Modern term structure theory and problems with traditional term structure theories.
Chapter 10		
Bond Prices and Yields	Straight bond prices and yield to maturity (YTM).	Calculate straight bond prices; calculate yield to maturity.
	The concept of duration and bond risk measures based on duration.	Calculate and interpret a bond's duration. The dolla value of an "01," and the yield value of a 32nd.
	Dedicated portfolios and reinvestment risk.	Learn how to create a dedicated portfolio and show its exposure to reinvestment risk.
	Immunization.	Minimize the uncertainty concerning the value of a bond portfolio at its target date.

Chapters	Selected Topics of Interest	Learning Outcome/Comment				
PART FOUR Portfolio Ma	anagement					
Chapter 11						
Diversification and Risky Asset Allocation	Expected returns and variances.	Calculating expected returns and variances using equal and unequal probabilities.				
	Portfolios and the effect of diversification on portfolio risk. Updated section: The fallacy of time diversification.	Compute portfolio weights, expected returns, variances, and why diversification works.				
	The importance of asset allocation.	The effect of correlation on the risk-return trade-off.				
	The Markowitz efficient frontier and illustrating the importance of asset allocation using three securities.	Compute risk-return combinations using various portfolio weights for three assets.				
Chapter 12						
Return, Risk, and the Security Market Line	Diversification, systematic and unsystematic risk.	Total risk is comprised of unsystematic and systematic risk and only unsystematic risk can be reduced through diversification.				
	The security market line and the reward- to-risk ratio.	The security market line describes how the market rewards risk. All assets will have the same reward- to-risk ratio in a competitive financial market.				
	Measuring systematic risk with beta. Calculating beta using regression.	The average beta is 1.00. Assets with a beta greater than 1.00 have more than average systematic risk.				
	The capital asset pricing model (CAPM).	Expected return depends on the amount and reward for bearing systematic risk as well as the pure time value of money.				
	Extending CAPM.	One of the most important extensions of the CAPM is the Fama-French three-factor model.				
Chapter 13						
Performance Evaluation and Risk Management	Performance evaluation measures. New material: The Sortino ratio.	Calculate and interpret the Sharpe ratio, the Sortino ratio, the Treynor ratio, and Jensen's alpha. Also, calculate alpha using regression, calculate an information ratio, and calculate a portfolio's <i>R</i> -squared.				
	Sharpe-optimal portfolios.	The portfolio with the highest possible Sharpe ratio given the assets comprising the portfolio is Sharpe optimal.				
	Value-at-Risk (VaR).	VaR is the evaluation of the probability of a significant loss.				
	Example showing how to calculate a Sharpe-optimal portfolio.	Combines the concepts of a Sharpe ratio, a Sharpe-optimal portfolio, and VaR.				
PART FIVE Futures and C	Options					
Chapter 14						
Futures Contracts	The basics of futures contracts and using them to hedge price risk. Detailed example: hedging an inventory using futures markets.	Futures quotes from the Internet and financial press, short and long hedging, futures accounts.				
	Spot-futures parity.	Basis, cash markets, and cash-futures arbitrage.				
	Stock index futures. <i>New example:</i> Changing the beta of a stock portfolio to zero using stock index futures.	Index arbitrage, speculating with stock index futures, and hedging stock market risk with stock index futures.				
	Hedging interest rate risk with futures.	We show how to use portfolio duration when deciding how many futures contracts to use to hedge a bond portfolio.				

Chapters	Selected Topics of Interest	Learning Outcome/Comment
Chapter 15		
Stock Options	Option basics and option price quotes. New material: Weekly options.	The difference between call and put options, European and American options, online option price quotes, and option chains.
	Option intrinsic value.	Know how to calculate this important aspect of option prices.
	Option payoffs and profits.	Diagram long and short option payoffs and profits for calls and puts.
	Using options to manage risk and option trading strategies. <i>Updated material:</i> Credit default swaps (CDSs).	Protective puts, covered calls, and straddles.
	Option pricing bounds and put-call parity.	Upper and lower pricing bounds for call and put options. Showing how a call option price equals a put option price, the price of an underlying share of stock, and appropriate borrowing.
PART SIX Topics in Inves	tments	
Chapter 16		
Option Valuation	The one-period and two-period binomial option pricing model.	How to compute option prices using this option pricing model—by hand and by using an online option calculator.
	The Black-Scholes option pricing model.	How to compute option prices using this famous option pricing model—by hand and by using an online option calculator.
	Measuring the impact of changes in option inputs.	Computing call and put option deltas.
	Hedging stock with stock options.	Using option deltas to decide how many option contracts are needed to protect a stock's price from feared declines in value.
	Employee stock options (ESOs) and their valuation.	Features of ESOs, repricing ESOs, and ESO valuation.
Chapter 17		
Projecting Cash Flow and Earnings	The basics of financial statements.	Income statement, balance sheet, cash flow statement, performance, and price ratios.
	Financial statement forecasting using the percentage of sales approach.	Preparing pro forma income statements and balance sheets to examine the potential amount of external financing needed.
	Updated material: A detailed case study valuing Starbucks Corporation.	Using actual financial data to prepare pro forma income statements and balance sheets using different sales growth scenarios.
Chapter 18 (new combination of chapters)		
Corporate and Government Bonds	Corporate bond basics, types of corporate bonds, and corporate bond indentures.	Become familiar with the basics of the various types of corporate bonds and their obligations.
	Callable bonds, putable bonds, convertible bonds, and protective covenants.	Bond seniority provisions, call provisions, make- whole call provisions, put provisions, conversion provisions, and protective covenants.
	Government bonds basics emphasizing U.S. government debt, federal government agency securities, and municipal bonds.	Details of U.S. Treasury bills, notes, bonds, STRIPS, agency bonds, and features of various types of municipal bonds.
	Bond credit ratings and junk bonds.	Assessing the credit quality of a bond issue.

Selected Topics of Interest	Learning Outcome/Comment	
New material: The process of top-down analysis.	Be able to funnel the choices of thousands of individual stocks through macroeconomic and industry filters.	
New material: Measure the level of economic activity globally and domestically.	Understand GDP, Real GDP, business cycles, economic indicators, and the effects of exchange rates on international investments.	
New material: Understand the relation of monetary and fiscal policies to economic activity.	The role of the Federal Reserve, money supply, and government policies on taxation.	
New material: Identify industry sensitivity to business cycles.	Identify the S&P sectors, compare companies within sectors, use Porter's five forces.	
Fixed-rate mortgages and prepayment.	Presents home mortgage principal and interest calculations.	
Secondary mortgage markets and reverse mortgages.	The function of GNMA and its clones, and the PSA mortgage prepayment model.	
Collateralized mortgage obligations, CMOs.	Describes how cash flows from mortgage pools are carved up and distributed to investors.	
	Selected Topics of InterestNew material: The process of top-down analysis.New material: Measure the level of economic activity globally and domestically.New material: Understand the relation of monetary and fiscal policies to economic activity.New material: Identify industry sensitivity to business cycles.Fixed-rate mortgages and prepayment.Secondary mortgage markets and reverse mortgages.Collateralized mortgage obligations, CMOs.	

Features

Pedagogical Features

Learning Objectives

You're probably going to be a mutual fund investor very soon, so you should definitely know the following:

1. The different

 How to find information about how mutual funds

types of mutual funds. 2. How mutual funds operate.

have performed.

 The workings of Exchange-Traded Funds (ETFs) and hedge funds.

From your feedback, we have included many pedagogical features in this text that will be valuable learning tools for your students. This walkthrough highlights some of the most important elements.

chapter 4

Chapter Openers

These one-paragraph introductions for each chapter present scenarios and common misconceptions that may surprise you. An explanation is more fully developed in the chapter.

Learning Objectives

Objectives next to the opener outline learning goals for the chapter.

CFA[™] Exam Map

This feature maps topics within each chapter to readings from the CFA[™] curriculum.

Mutual Funds and Other Investment Companies

"Take calculated risks. That is quite different from being rash."

With only \$2,000 to invest, you can easily own shares in Microsoft, GM, McDonald's, IBM, Coke, and many more stocks through a mutual fund. Or, you can invest in a portfolio of government bonds or other investments. Indeed, many thousands of different mutual funds are available to investors. In fact, there are about as many mutual funds as there are different stocks traded on the NASDAQ and the New York Stock Exchange combined. There are funds for aggressive investors, conservative investors, short-term investors, and long-term investors. There are bond funds, stock funds, international funds, and youname-it funds. Is there a right fund for you? This chapter will help you find out.

As we discussed in an earlier chapter, if you do not wish to actively buy and sell individual securities on your own, you can invest in stocks, bonds, or other financial assets through a *mutual fund*. Mutual funds are simply a means of combining or pooling the funds of a large group of investors. The buy and sell decisions for the resulting pool are then made by a fund manager, who is compensated for the service provided.

Because mutual funds provide indirect access to financial markets for individual investors, they are a form of financial intermediary. In fact, mutual funds are now the largest type of intermediary in the United States, followed by commercial banks and life insurance companies.

CFA™ Exam Topics in This Chapter:

- 1 Discounted cash flow applications (L1, S2)
- 2 Alternative investments (L1, S18)
- 3 Soft dollar standards (L2, S1)
- 4 Alternative investments portfolio management (L3, S13)

Go to www.mhhe.com/jmd7e for a guide that aligns your textbook with CFA readings.

Check This

Every major section in each chapter ends with questions for review. This feature helps students test their understanding of the material before moving on to the next section.



4.1a What are some advantages of investing in mutual funds?

4.1b What are some drawbacks of investing in mutual funds?

risk-free rate The rate of return on a riskless risk. Thus, we will call the rate of return on such debt the **risk-free rate**, and we will use it as a kind of investing benchmark. A particularly interesting comparison involves the virtually risk-free return on T-bills and

the risky return on common stocks. The difference between these two returns can be interpreted as a measure of the *excess return* on the average risky asset (assuming that the stock of a large U.S. corporation has about average risk compared to all risky assets).

Key Terms

Key terms are indicated in bold and defined in the margin. The running glossary in the margin helps students quickly review the basic terminology for the chapter.

Web Addresses -

Web sites are called out in the margin, along with a notation of how they relate to the chapter material.

Want to have a career in financial advice? See www.cfainstitute.org

financial advice? See www.cfainstitute.org and www.cfp.net RESOURCES Probably the most obvious constraint, and the one to which many students can most easily relate, is *resources*. Obviously, if you have no money, you cannot invest at all. Beyond that,

ous constraints. We discuss five of the most common and important constraints next

certain types of investments and investment strategies generally have minimum requirements. What is the minimum resource level needed? The answer to this question depends on the investment strategy, so there is no precise answer. Through mutual funds, initial investments in the stock market can be made for as little as \$250, with subsequent investments as small as \$50 or less. However, because minimum commission levels, account fees, and other costs

INVESTMENT UPDATES

BUFFETT ON TAXES AND TRADING

Through my favorite comic strip, "Li'l Abner," I got a chance during my youth to see the benefits of delayed taxes, though I missed the lesson at the time. Making his readers feel superior, Li'l Abner bungled happily, but moronically, through life in Dogpatch. At one point he became infatuated with a New York temptress, Appassionatta Van Climax, but despaired of marrying her because he had only a single silver dollar and she was interested solely in millionaires. Dejected, Abner took his problem to Old Man Mose, the font of all knowledge in Dogpatch. Said the sage: Double your money 20 times and Appassionatta will be yours (1, 2, 4, 8, ..., 1,048,576).

My last memory of the strip is Abner entering a roadhouse, dropping his dollar into a slot machine, and hitting a jackpot that spilled money all over the floor. Meticulously following Mose's advice, Abner picked up two 20 years only have accumulated \$22,370. Indeed, had he kept on both getting his annual doubles and paying a 35% tax on each, he would have needed 7% years more to reach the \$1 million required to win Appassionatta.

But what if Abner had instead put his dollar in a single investment and held it until it doubled the same 27% times? In that case, he would have realized about \$200 million pretax or, after paying a \$70 million tax in the final year, about \$130 million after-tax. For that, Appassionatta would have crawled to Dogpath. Of course, with 27% years having passed, how Appassionatta would have looked to a fellow sitting on \$130 million is another question.

What this little tale tells us is that tax-paying investors will realize a far, far greater sum from a single investment that compounds internally at a given rate than from a succession of investments compounding at the same

Investment Updates

These boxed readings, reprinted from various business press sources, provide additional real-world events and examples to illustrate the material in the chapter. Some articles from the past two years highlight very recent events, and others present events of more historical significance.

Work the Web

Various screenshots appear throughout the text. These exercises illustrate how to access specific features of selected Web sites in order to expand students' knowledge of current investment topics.

WORK THE WEB

You can find the short interest for the current month in many financial publications. But what if you want a longer history of the shares sold short for a particular company? At www.nasdaq.com, you can find the short interest for companies listed on the NASDAQ for the previous 11 months. We went to the site and looked up Yahoo!, and here is what we found:

Settlement Date	Short Interest	Avg Daily Share Volume	Days To Cover
4/13/2012	29,292,480	14,279,539	2.051360
3/30/2012	31,350,687	16,248,381	1.929465
3/15/2012	28,440,779	12,823,105	2.217932
2/29/2012	27,996,626	19,127,092	1.463716
2/15/2012	26,366,458	22,728,324	1.160070
1/31/2012	31,840,444	18,429,404	1.727698
1/13/2012	42,012,870	16,781,396	2.503538
12/30/2011	36,650,629	22,027,145	1.663885
12/15/2011	36,963,198	21,063,241	1.754868

As you can see, the short interest in Yahoo! fell from about 37 million shares in December 2011 to about 29 million shares in April 2012. Why would you want a history of short sales? Some investors use short sales as a technical indicator, which we discuss in a later chapter. Here's a question for you: What do you think "Days to Cover" means? It is the ratio of short interest to average daily share volume. Thus, "Days to Cover" measures how many days of normal trading would be necessary to completely cover all outstanding short interest.



Numbered Examples

Example boxes are integrated throughout the chapters to reinforce the content and demonstrate to students how to apply what they've learned. Each example displays an intuitive or mathematical application in a step-by-step format. There is enough detail in the explanations so that the student doesn't have to look elsewhere for additional information.

Spreadsheet Analysis

Self-contained spreadsheet examples show students how to set up spreadsheets to solve problems—a vital part of every business student's education.

SPREADSHEET ANALYSIS

Using a Spreadsheet to Calculate Average Returns and Volatilities Here is an Excel spreadsheet summarizing the formulas and analysis needed to calculate average returns and standard deviations using the 1990s as an example:

	А	В	С	D	E	F	G	Н
1								
2	l	Using a spreadsheet to calculate average returns and standard deviation				IS		
3								
4	Looking back in the chapter, the data suggest that the 1990s were one							
5	of the best	decades fo	r stock marke	t investors.	We will find	out just hov	v good by	
6	calculating	calculating the average returns and standard deviations for this period. Here are the						
7	year-by-year returns on the large-company stocks:							
8								
9		Year	Return (%)	Year	Return (%)			
10		1990	-3.10	1995	37.58			
11		1991	30.46	1996	22.96			
12		1992	7.62	1997	33.36			
13		1993	10.08	1998	28.58			
14		1994	1.32	1999	21.04			
15								
16	Average return (%): 18.99							
17		Standard d	eviation (%):	14.16				

Numbered Equations

Key equations are highlighted and numbered sequentially. For easy reference, an appendix at the end of the book lists these key equations by chapter. In our example, the price at the beginning of the year was \$50 per share and the dividend paid during the year on each share was \$.40. If we divide the dividend by the beginning stock price, the result is the **dividend yield**:

Dividend yield =
$$D_{t+1} / P_t$$
 (1.1)
= \$.40 / \$50 = .0080 = 0.80%

This calculation says that for each dollar we invested we received 80 cents in dividends. The second component of our percentage return is the **capital gains yield**. This yield is calculated as the change in the price during the year (the capital gain) divided by the beginning price. With the case 1 ending price, we get:

c

apital gains yield =
$$(P_{t+1} - P_t) / P_t$$
 (1.2)
= (\$55.60 - \$50.00) / \$50.00
= \$5.60 / \$50 = .1120 = 11.20%

This 11.20 percent yield means that for each dollar invested we got about 11 cents in capital gains (HOG heaven).

Putting it all together, per dollar invested, we get 80 cents in dividends and \$11.20 in capital gains for a total of \$12.00. Our **total percent return** is 12 cents on the dollar, or 12.00 percent. When a return is expressed on a percentage basis, we often refer to it as the *rate of return*, or iust "return," on the investment. Notice that if we combine the formulas for



Figures and Tables

This text makes extensive use of real data and presents them in various figures and tables. Explanations in the narrative, examples, and end-of-chapter problems refer to many of these exhibits.

2.6 Summary and Conclusions

In this chapter, we cover many aspects of the investing process—which we summarize by the chapter's important concepts.

- 1. The importance of an investment policy statement.
- A. The investment policy statement (IPS) identifies the objectives (risk and return) of an investor, as well as the constraints the investor faces in achieving these objectives.
- B. The IPS provides an investing "road map" and will influence the strategies, type of account, and holdings an investor chooses.
- 2. The various types of securities brokers and brokerage accounts.
- A. Opening a brokerage account is straightforward and really much like opening a bank account. You supply information and sign agreements with your broker. Then you write a check and provide instructions on how you want your money invested.
- B. Brokers are traditionally divided into three groups: full-service brokers, discount brokers, and deep-discount brokers. What distinguishes the three groups is the level of service they provide and the resulting commissions they charge. In recent years, the boundaries among the groups have blurred.
- C. Your broker does not have a duty to provide you with guaranteed purchase and sale recommendations. However, your broker does have a duty to exercise reasonable care in formulating recommendations. Your broker has a legal duty to act in your best

Summary and Conclusions —

Each chapter ends with a summary that highlights the important points of the chapter. This material provides a handy checklist for students when they review the chapter.

Getting Down to Business

For instructors looking to give their students a taste of what it means to be an investment manager, this feature (at the end of each chapter) acts as a first step by explaining to students how to apply the material they just learned. The Getting Down to Business boxes encourage studentswhether for practice in a trading simulation, or with real money-to make investment decisions, and they also give some helpful tips to keep in mind. These boxes include a QR code link to a handy Weblog written by the authors.

GETTING DOWN TO BUSINESS

This chapter covered the basics of policy statements, brokerage accounts, some important trade types, and, finally, some big-picture issues regarding investment strategies. How should you, as an investor or investment manager, put this information to work?

The answer is that you need to open a brokerage account! Investing is like many activities; the best way to learn is by making mistakes. Unfortunately, making mistakes

activities; the best way to learn is by making mistakes. Unfortunately, making mistakes with real money is an expensive way to learn, so we don't recommend trying things like short sales with real money, at least not at first. Instead, to learn about how to trade and gain some experience with making (and losing) money, you should open a Stock-Trak account (or a similar simulated brokerage account). Take it seriously. Try various trade types and strategies and see how they turn out. The important thing to do is to follow your trades and try to understand why you made or lost money and also why you made or lost the amount you did.

In a similar vein, you should carefully review your account statements to make sure you understand exactly what each item means and how your account equity is calculated.

After you have gained some experience trading "on paper," you should open a real account as soon as you can pull together enough money. Try visiting some online bookers such as DoAmeritade to find out the minimum amount you need to open an account. The amount has been declining. In 2012, you could open a cash account for as little as \$500, but to open a margin account, you need in the area of \$2,000. Or, you can visit www.sharebuilder.com and www.buvandhold.com to open accounts with no money at all!

Looking back at Chapter 1, you know that it's important to get started early. Once you have a real account, however, it's still a good idea to keep a separate "play money" account to test trading ideas to make sure you really understand them before committing your precious real money.

Chapter Review Problems and Self-Test

- 1. Front-End Loads (CFA2) The Madura HiGro Fund has a net asset value of \$50 per share. It charges a 3 percent load. How much will you pay for 100 shares?
- 2. Turnover (CFA2) The Starks Income Fund's average daily total assets were \$100 million for the year just completed. Its stock purchases for the year were \$20 million, while its sales were \$12.5 million. What was its turnover?

Answers to Self-Test Problems

1. You will pay 100 times the offering price. Since the load is computed as a percentage of the offering price, we can compute the offering price as follows:

Net asset value = $(1 - \text{Front-end load}) \times \text{Offering price}$

- In other words, the NAV is 97 percent of the offering price. Since the NAV is \$50, the offering price is \$50/.97 = \$51.55. You will pay \$5,155 in all, of which \$155 is a load.
- 2. Turnover is the lesser of purchases or sales divided by average daily assets. In this case, sales are smaller at \$12.5, so turnover is 12.5/100 = .125 times.

Chapter Review Problems and Self-Test

Students are provided with one to three practice problems per chapter with worked-out solutions to test their abilities in solving key problems related to the content of the chapter.

Test Your Investment Quotient

An average of 15 multiple-choice questions are included for each chapter, many of which are taken from past CFA exams. This text is unique in that it presents CFA questions in multiplechoice format-which is how they appear on the actual exam. Answers to these questions appear in Appendix A.

IQ

nte vieit ue at

Test Your Investment Quotient 1. Prices and Returns (LO1, CFA1) You plan to buy a common stock and hold it for one year You expect to receive both \$1.50 from dividends and \$26 from the sale of the stock at the end of the year. If you wanted to earn a 15 percent rate of return, what is the maximum price you would pay for the stock today? a. \$22.61b. \$23.91 c. \$24.50 **d.** \$27.50 2. Returns (LO1, CFA1) A portfolio of non-dividend-paying stocks earned a geometric mean return of 5 percent between January 1, 2004, and December 31, 2010. The arithmetic mean re-turn for the same period was 6 percent. If the market value of the portfolio at the beginning of 2004 was \$100,000, the market value of the portfolio at the end of 2010 was closest to: a. \$135,000b. \$140,710 c. \$142,000d. \$150,363 3. Standard Deviation (LO4, CFA2) Which of the following statements about standard deviation is true? Standard devia

- a. Is the square of the variance
- b. Can be a positive or negative number.
- c. Is denominated in the same units as the original data.
- d. Is the arithmetic mean of the squared deviations from the mean.

4. Normal Distribution (LO4, CFA3) An investment strategy has an expected return of 12 percent and a standard deviation of 10 percent. If the investment returns are normally distributed, the probability of earning a return less than 2 percent is closest to:

Concept Questions

- 1. Margin (LO3, CFA4) What does it mean to purchase a security on margin? Why might you do it?
- 2. Short Sales (LO4, CFA5) What does it mean to sell a security short? Why might you do it?
- Margin Requirements (LO3, CFA4) What is the reason margin requirements exist?
 Allocation versus Selection (LO1, CFA2) What is the difference between asset allocation
- and security selection?

Concept Questions

At the end of every chapter are 10 to 15 concept questions that further reinforce key concepts found throughout the chapter.

Questions and Problems

A variety of problems (average of 20 per chapter) are included in each chapter to test students' understanding of the conceptual and mathematical elements. Each problem is labeled with the subject and the level—core or intermediate. Selected answers appear in Appendix B, and complete solutions are included in the Instructor Web site.

Questions and Problems

Core Ouestions

 Calculating Margin (LO3, CFA4) Carson Corporation stock sells for \$17 per share, and you've decided to purchase as many shares as you possibly can. You have \$31,000 available to invest. What is the maximum number of shares you can buy if the initial margin is 60 percent?
 Margin (LO3, CFA4) You purchase 275 shares of 2nd Chanec Co. stock on margin at a price of \$353, Your broker requires you to deposit \$8,000. What is your margin leam annum? What is a

CFA Exam Review by Schweser

[CFA1, CFA7, CFA10, CFA11]

Barbara Analee, a registered nurse and businesswoman, recently retired at age 50 to pursue a life as a blues singer. She had been running a successful cosmetics and aesthetics business. She is married to form, a retired scientist (age 55). They have saved 35 million in their portfolio and now they want to travel the world. Their three children are all grown and out of college and have begun their own families. Barbara now has two grandchildren. Barbara and Tom feel that they have achieved a comfortable portfolio level to support their family's needs for the foresceable future.

To meet their basic living expenses, Tom and Barbara feel they need \$75,000 per year in today's dollars (before taxes) to live comfortably. As a trained professional, Barbara likes to be actively involved in intensively researching investment opportunities. Barbarara and Tom want to be able to provide \$10,000 per year (pretax) indexed for inflation to each of their grandchildren over the next 10 years for their college education. They also want to set aside \$15,000 each year (pretax) indexed for inflation for traveling for her musical performances around the United States. They have no debt. Most of their portfolio is currently in large-cap U.S. stocks and Treasury notes.

They have approached Pamela Jaycoo, CFA, for guidance on how to best achieve their financial goals. Inflation is expected to increase at an annual rate of 3 percent into the foreseeable future.

1. What is the Analee's return objective?

- a. 6.67 percent
- **b.** 6.17 percent **c.** 3.83 percent
- What is their tolerance for risk?
- a. Average
- b. Below average
- c. Above average

CFA Exam Review by Schweser

Unique to this text! These reviews are excerpted from Schweser, a leader in CFA exam preparation. Each review addresses chapter content but in a way that is consistent with the format of the actual CFA exam.

What's on the Web?

These end-of-chapter activities show students how to use and learn from the vast amount of financial resources available on the Internet.

What's on the Web?

- Risk Tolerance As we discussed in the chapter, risk tolerance is based on an individual's
 personality and investment goals. There are numerous risk tolerance questionnaires on the Web.
 One, provided by Merrill Lynch, is located at individual.ml.com. Go to the Web site, locate the
 questionnaire, and take the quiz. How conservative or aggressive are you?
- 2. Short Interest You can find the number of short sales on a particular stock at <u>finance, yahoo.com</u>. Go to the site and find the number of shares short sold for ExxonMobil (XOM) under the "Key Statistics" link. How many shares are sold short in the current month? What about the previous month? What do the "Percent of Float" and "Short Ratio" mean?
- 3. Broker Call Money Rate What is the current broker call money rate? To find out, go to





To access the Stock-Trak Exercise for this chapter, please visit the book Web site at www.mhhe.com/jmd7e and choose the corresponding chapter.

Stock-Trak Exercises

Unique to this text! This text is the only book that incorporates Stock-Trak Portfolio Simulations® exercises. Stock-Trak is one of the most successful trading simulations with over 30,000 college students having trading accounts each semester (see Supplements for more information). Go to the next level in teaching your students about investments management by encouraging your students to use this product. Chapters with Stock-Trak Exercises have the logo and the URL for the book's Web site. The actual exercise and questions related to the chapter are presented in both the Student and Instructor portions of the Web site. Instructors and students must be registered for Stock-Trak in order to make trades (see the Supplement section of the Preface or the insert card for more information).

Resources

Teaching and Learning Supplements

We have developed a number of supplements for both teaching and learning to accompany this text. Each product has been significantly revised for the seventh edition.

Digital Solutions

Online Learning Center (OLC):

Online Support at www.mhhe.com/jmd7e

The Online Learning Center (OLC) contains access to additional Web-based study and teaching aids created for this text, such as:

Student Support

Student-Narrated PowerPoints created by Lynn Kugele, University of Mississippi

Students all learn differently and these chapter PowerPoints were created with that rationale in mind. The interactive presentations provide detailed examples demonstrating how to solve key problems from the text. The slides are accompanied by an audio narration. They can be purchased as part of the premium content package available for \$10 and then viewed online.

Excel Templates

Corresponding to most end-of-chapter problems, each template allows the student to work through the problem using Excel, reinforcing each concept. Each end-of-chapter problem with a template is indicated by an Excel icon in the margin beside it.

Self-Study Chapter Quizzes

Quizzes consist of 10–15 multiple-choice questions on various chapter topics. They reveal a score instantly and provide feedback to help students study.

Other Features

Be sure to check out the other helpful features found on the OLC including key-term flashcards, helpful Web links, and more!

Instructor Support

The Instructor's Edition of the OLC contains the following assets:

PowerPoint Presentation, *prepared by Thomas W. Miller Jr.*, Mississippi State University

This product, created by one of the authors, contains over 300 slides with lecture outlines, examples, and images and tables from the text.

Instructor's Manual, prepared by Steven D. Dolvin, CFA, Butler University

Developed by one of the authors, the goals of this product are to outline chapter material clearly and provide extra teaching support. The first section of the Instructor's Manual includes an annotated outline of each chapter with suggested Web sites, references to PowerPoint slides, teaching tips, additional examples, and current events references.

Solutions Manual, prepared by Steven D. Dolvin, CFA, Butler University

The Solutions Manual contains the complete worked-out solutions for the end-ofchapter questions and problems.

Test Bank, prepared by Lynn Kugele, University of Mississippi

With almost 1,500 questions, this Test Bank, in Microsoft Word, provides a variety of question formats (true-false, multiple-choice, fill-in-the-blank, and problems) and levels of difficulty to meet any instructor's testing needs.

Computerized Test Bank (Windows)

This computerized version of the Test Bank utilizes McGraw-Hill's EZ Test testing software to quickly create customized exams. This user-friendly program allows instructors to sort questions by format; edit existing questions or add new ones; and scramble questions for multiple versions of the same test.

Additional Resources Packaged with Your New Text

Stock-Trak Portfolio Simulation

Give your students investment management experience! McGraw-Hill/Irwin has partnered with *Stock-Trak* and is providing a **free** subscription to the *Stock-Trak Portfolio Simulation* for one semester with the purchase of every new copy of *Fundamentals of Investments: Valuation and Management, Seventh Edition* by Jordan, Miller, and Dolvin. *Stock-Trak* gives students \$1,000,000 and allows them to trade stocks, options, futures, bonds, mutual funds, and international stocks—no other simulation offers all these types of securities! More than 600 professors have used this service, and around 30,000 college students each semester participate. All trades are done on the Web at www.stocktrak.com. See this site for more information or use the Stock-Trak card bound into this text. Stock-Trak exercises are available on the book Web site, www.mhhe.com/jmd7e.

McGraw-Hill Connect Finance

Less Managing. More Teaching. Greater Learning.

McGraw-Hill *Connect Finance* is an online assignment and assessment solution that connects students with the tools and resources they'll need to achieve success. *Connect* helps prepare students for their future by enabling faster learning, more efficient studying, and higher retention of knowledge.

McGraw-Hill Connect Finance Features

Connect Finance offers a number of powerful tools and features to make managing assignments easier, so faculty can spend more time teaching. With *Connect Finance*, students can engage with their coursework anytime and anywhere, making the learning process more accessible and efficient. *Connect Finance* offers you the features described below.

Simple Assignment Management With *Connect Finance*, creating assignments is easier than ever, so you can spend more time teaching and less time managing. The assignment management function enables you to:

- Create and deliver assignments easily with selectable end-of-chapter questions and test bank items.
- Streamline lesson planning, student progress reporting, and assignment grading to make classroom management more efficient than ever.
- Go paperless with the eBook and online submission and grading of student assignments.

Smart Grading When it comes to studying, time is precious. *Connect Finance* helps students learn more efficiently by providing feedback and practice material when they need it, where they need it. When it comes to teaching, your time is also precious. The grading function enables you to:

- Have assignments scored automatically, giving students immediate feedback on their work and side-by-side comparisons with correct answers.
- Access and review each response; manually change grades or leave comments for students to review.
- Reinforce classroom concepts with practice tests and instant quizzes.

Instructor Library The *Connect Finance* Instructor Library is your repository for additional resources to improve student engagement in and out of class. You can select and use any asset that enhances your lecture.

Student Study Center The *Connect Finance* Student Study Center is the place for students to access additional resources. The Student Study Center:

- Offers students quick access to lectures, practice materials, eBooks, and more.
- Provides instant practice material and study questions, easily accessible on the go.

Student Progress Tracking *Connect Finance* keeps instructors informed about how each student, section, and class is performing, allowing for more productive use of lecture and office hours. The progress-tracking function enables you to:

- View scored work immediately and track individual or group performance with assignment and grade reports.
- Access an instant view of student or class performance relative to learning objectives.

Lecture Capture through Tegrity Campus For an additional charge, Lecture Capture offers new ways for students to focus on the in-class discussion, knowing they can revisit important topics later. This can be delivered through *Connect* or separately. See below for more details.

McGraw-Hill Connect Plus Finance

McGraw-Hill reinvents the textbook learning experience for the modern student with *Connect Plus Finance*. A seamless integration of an eBook and *Connect Finance*, *Connect Plus Finance* provides all of the *Connect Finance* features plus the following:

- An integrated eBook, allowing for anytime, anywhere access to the textbook.
- Dynamic links between the problems or questions you assign to your students and the location in the eBook where that problem or question is covered.
- A powerful search function to pinpoint and connect key concepts in a snap.

In short, *Connect Finance* offers you and your students powerful tools and features that optimize your time and energies, enabling you to focus on course content, teaching, and student learning. *Connect Finance* also offers a wealth of content resources for both instructors and students. This state-of-the-art, thoroughly tested system supports you in preparing students for the world that awaits.

For more information about *Connect*, go to **www.mcgrawhillconnect.com**, or contact your local McGraw-Hill sales representative.

Tegrity Campus: Lectures 24/7

Tegrity Campus is a service that makes class time available 24/7 by automatically capturing every lecture in a searchable format for students to review when they study and complete assignments. With a simple one-click start-and-stop process, you capture all computer screens and corresponding audio. Students can replay any part of any class with easy-to-use browser-based viewing on a PC or Mac.

Educators know that the more students can see, hear, and experience class resources, the better they learn. In fact, studies prove it. With Tegrity Campus, students quickly recall key moments by using Tegrity Campus's unique search feature. This search helps students efficiently find what they need, when they need it, across an entire semester of class recordings. Help turn all your students' study time into learning moments immediately supported by your lecture.

To learn more about Tegrity watch a two-minute Flash demo at http://tegritycampus.mhhe.com.

McGraw-Hill Customer Care Contact Information

At McGraw-Hill, we understand that getting the most from new technology can be challenging. That's why our services don't stop after you purchase our products. You can e-mail our Product Specialists 24 hours a day to get product training online. Or you can search our knowledge bank of Frequently Asked Questions on our support Web site. For Customer Support, call **800-331-5094**, or visit **www.mhhe.com/support**. One of our Technical Support Analysts will be able to assist you in a timely fashion.

Brief Contents

PART ONE Introduction 1

1. A Brief History of Risk and Return 1

- 2. The Investment Process 41
- **3.** Overview of Security Types 77
- 4. Mutual Funds and Other Investment Companies 102

PART TWO Stock Markets 144

- 5. The Stock Market 144
- 6. Common Stock Valuation 179
- 7. Stock Price Behavior and Market Efficiency 221
- 8. Behavioral Finance and the Psychology of Investing 257

PART THREE Interest Rates and Bond Valuation 298

- 9. Interest Rates 298
- 10. Bond Prices and Yields 337

PART FOUR Portfolio Management 372

- 11. Diversification and Risky Asset Allocation 372
- 12. Return, Risk, and the Security Market Line 406
- **13.** Performance Evaluation and Risk Management 441

PART FIVE Futures and Options 468

- **14.** Futures Contracts 468
- **15.** Stock Options 499

PART SIX Topics in Investments 540

- 16. Option Valuation 540
- 17. Projecting Cash Flow and Earnings 578
- **18.** Corporate and Government Bonds 615
- **19.** Global Economic Activity and Industry Analysis 659
- 20. Mortgage-Backed Securities (Web site only) 20-1

APPENDIXES

- A Answers to Test Your Investment Quotient Questions 682
- Answers to Selected Questions and Problems 686
- C Key Equations 688

Index 692

Contents

Preface vii

PART ONE

- Introduction 1 1. A Brief History of Risk and Return 1 1.1 Returns 2 Dollar Returns 2 Percentage Returns 4 A Note on Annualizing Returns 6 1.2 The Historical Record 7 A First Look 8 A Longer Range Look 9 A Closer Look 9 2008: The Bear Growled and Investors Howled 12 1.3 Average Returns: The First Lesson 14 Calculating Average Returns 15 Average Returns: The Historical Record 15 **Risk Premiums** 15 The First Lesson 18 1.4 Return Variability: The Second Lesson 18 Frequency Distributions and Variability 18 The Historical Variance and Standard Deviation 19 The Historical Record 21
 - Normal Distribution 22 The Second Lesson 23 1.5 More on Average Returns 26
 - Arithmetic versus Geometric Averages 26 Calculating Geometric Average Returns 26 Arithmetic Average Return or Geometric Average Return? 28 Dollar-Weighted Average Returns 29
 - 1.6 Risk and Return 31 The Risk-Return Trade-Off 31 A Look Ahead 32
 - Summary and Conclusions 32 1.7

- 2. The Investment Process 41 2.1 The Investment Policy Statement 42 Objectives: Risk and Return 42 **Investor Constraints** 42 Strategies and Policies 47 2.2 Investment Professionals 49 Choosing a Broker/Advisor 49 **Online Brokers** 50 Investor Protection 50 Broker-Customer Relations 51 2.3 Types of Accounts 52 Cash Accounts 52 Margin Accounts 52 Annualizing Returns on a Margin Purchase 56 Hypothecation and Street Name Registration 57 Retirement Accounts 58 2.4 Types of Positions 59 Basics of a Short Sale 59 Short Sales: Some Details 60 Short-Sale Constraints 63 2.5 Forming an Investment Portfolio 65 Some Risk Tolerance Scores 65 Risk and Return 65 Investor Constraints 66 Strategies and Policies 66 More on Asset Allocation 67
 - REITs 67
 - Summary and Conclusions 68 2.6
- 3. Overview of Security Types 77
 - **Classifying Securities** 78 3.1
 - 3.2 Interest-Bearing Assets 78 Money Market Instruments 78 Fixed-Income Securities 79

3.3	Equities 82
	Common Stock 82
	Preferred Stock 82
	Common Stock Price Quotes 84
3.4	Derivatives 87
	Futures Contracts 87
	Futures Price Quotes 88
	Gains and Losses on Futures Contracts 90
3.5	Option Contracts 90
	Option Terminology 90
	Options versus Futures 91
	Option Price Quotes 91
	Gains and Losses on Option Contracts 93
	Investing in Stocks versus Options 93
3.6	Summary and Conclusions 94
Mut Con	ual Funds and Other Investment Ipanies 102
4.1	Advantages and Drawbacks of Mutual
	Fund Investing 103
	Advantages 103
	Drawbacks 104
4.2	Investment Companies and Fund Types 104
	Open-End versus Closed-End Funds 104
	Net Asset Value 105
4.3	Mutual Fund Operations 106
	Mutual Fund Organization and Creation 106
	Taxation of Investment Companies 107
	The Fund Prospectus and Annual Report 107
4.4	Mutual Fund Costs and Fees 107
	Types of Expenses and Fees 107
	Expense Reporting 109
	Why Pay Loads and Fees? 111

4.

- 4.5 Short-Term Funds 111 Money Market Mutual Funds 112 Money Market Deposit Accounts 114
- 4.6 Long-Term Funds 114
 Stock Funds 114
 Taxable and Municipal Bond Funds 116
 Stock and Bond Funds 118
 Mutual Fund Objectives: Recent Developments 119
- 4.7 Mutual Fund Performance 122 Mutual Fund Performance Information 122 How Useful Are Fund Performance Ratings? 123

- 4.8 Closed-End Funds, Exchange-Traded Funds, and Hedge Funds 125
 Closed-End Funds Performance Information 125
 The Closed-End Fund Discount Mystery 126
 Exchange-Traded Funds 127
 Hedge Funds 133
- 4.9 Summary and Conclusions 136

PART TWO Stock Markets 144

- 5. The Stock Market 144
 - 5.1 Private Equity versus Selling Securities to the Public 145
 Private Equity 145
 The Structure of Private Equity Funds 145
 Types of Private Equity Funds 146
 Selling Securities to the Public 147
 The Primary Market for Common Stock 147
 The Secondary Market for Common Stock 150
 Dealers and Brokers 150
 5.2 The New York Stock Exchange 153
 - NYSE Membership History 153
 Designated Market Makers 153
 Other NYSE Participants 154
 The NYSE Hybrid Market 154
 NYSE-Listed Stocks 155
 - 5.3 Operation of the New York Stock Exchange 155NYSE Floor Activity 156Special Order Types 157
 - 5.4 NASDAQ 160NASDAQ Operations 161NASDAQ Participants 161
 - 5.5 NYSE and NASDAQ Competitors 162
 - 5.6 Stock Market Information 164 The Dow Jones Industrial Average 164 Stock Market Indexes 164 More on Price-Weighted Indexes 168 The Dow Jones Divisors 169 More on Index Formation: Base-Year Values 169
 - 5.7 Summary and Conclusions 170

6. Common Stock Valuation 179

Models 193

- 6.1 Security Analysis: Be Careful Out There 180
- 6.2 The Dividend Discount Model 180 Constant Perpetual Growth 181 Historical Growth Rates 183 The Sustainable Growth Rate 185 Analyzing ROE 186
- 6.3 The Two-Stage Dividend Growth Model 188 Nonconstant Growth in the First Stage 190 The H-Model 192 Discount Rates for Dividend Discount Models 192 Observations on Dividend Discount
- 6.4 The Residual Income Model 193 Residual Income 194 The RIM versus the Constant Growth DDM 194
- 6.5 The Free Cash Flow Model 196Free Cash Flow 196The FCF Model versus the Constant Growth DDM 197
- 6.6 Price Ratio Analysis 199
 Price-Earnings Ratios 199
 Price-Cash Flow Ratios 200
 Price-Sales Ratios 200
 Price-Book Ratios 200
 Applications of Price Ratio Analysis 201
 Enterprise Value Ratios 202
- 6.7 An Analysis of the Procter & Gamble Company 203
 Using the Dividend Discount Model 205
 Using the Residual Income Model 205
 Using Price Ratio Analysis 207
- 6.8 Summary and Conclusions 209

7. Stock Price Behavior and Market Efficiency 221

- 7.1 Introduction to Market Efficiency 222
- 7.2 What Does "Beat the Market" Mean? 222
- 7.3 Foundations of Market Efficiency 222
- 7.4 Forms of Market Efficiency 223
- 7.5 Why Would a Market Be Efficient? 224
- 7.6 Some Implications of Market Efficiency 225
 Does Old Information Help Predict
 Future Stock Prices? 225
 Random Walks and Stock Prices 225

How Does New Information Get into Stock Prices? 226 Event Studies 226

- 7.7 Informed Traders and Insider Trading 229 Informed Trading 229 Insider Trading 229
- 7.8 How Efficient Are Markets? 231 Are Financial Markets Efficient? 231 Some Implications of Market Efficiency 233
- 7.9 Market Efficiency and the Performance of Professional Money Managers 234
- 7.10 Anomalies 237

 The Day-of-the-Week Effect 237
 The Amazing January Effect 237
 Turn-of-the-Year Effect 240
 Turn-of-the-Month Effect 240
 The Earnings Announcement Puzzle 241
 The Price-Earnings (P/E) Puzzle 241
- 7.11 Bubbles and Crashes 241

 The Crash of 1929 241
 The Crash of October 1987 242
 The Asian Crash 245
 The "Dot-Com" Bubble and Crash 246
 The Crash of October 2008 246
- 7.12 Summary and Conclusions 248
- Behavioral Finance and the Psychology of Investing 257
 - 8.1 Introduction to Behavioral Finance 258
 - 8.2 Prospect Theory 258
 Frame Dependence 259
 Loss Aversion 260
 Mental Accounting and House Money 261
 - 8.3 Overconfidence 262

 Overconfidence and Trading
 Frequency 262
 Overtrading and Gender: "It's (Basically)
 a Guy Thing" 263
 What Is a Diversified Portfolio to the
 Everyday Investor? 263
 Illusion of Knowledge 263
 Snakebite Effect 264

 8.4 Misperceiving Randomness and
 - Whispercerving Randonmess and Overreacting to Chance Events 265
 The "Hot-Hand" Fallacy 267
 The Gambler's Fallacy 268

- 8.5 More on Behavioral Finance 269 Heuristics 269 Herding 269 How Do We Overcome Bias? 269
- 8.6 Sentiment-Based Risk and Limits to Arbitrage 270 Limits to Arbitrage 271 The 3Com/Palm Mispricing 272 The Royal Dutch/Shell Price Ratio 272
- 8.7 Technical Analysis 274
 Why Does Technical Analysis Continue to Thrive? 274
 Dow Theory 275
 Elliott Waves 275
 Support and Resistance Levels 276
 Technical Indicators 276
 Relative Strength Charts 279
 Charting 279
 Fibonacci Numbers 285
 Other Technical Indicators 285
 8.8 Summary and Conclusions 287

PART THREE Interest Rates and Bond Valuation 298

- 9. Interest Rates 298
 - 9.1 Interest Rate History and Money Market Rates 299
 Interest Rate History 299
 Money Market Rates 301
 - 9.2 Money Market Prices and Rates 305
 Bank Discount Rate Quotes 305
 Treasury Bill Quotes 306
 Bank Discount Yields versus Bond
 Equivalent Yields 308
 Bond Equivalent Yields, APRs, and EARs 309
 - 9.3 Rates and Yields on Fixed-Income Securities 311 The Treasury Yield Curve 312 Rates on Other Fixed-Income Investments 313
 - 9.4 The Term Structure of Interest Rates 316 Treasury STRIPS 316 Yields for U.S. Treasury STRIPS 317
 - 9.5 Nominal versus Real Interest Rates 319 Real Interest Rates 319

The Fisher Hypothesis 320 Inflation-Indexed Treasury Securities 320

- 9.6 Traditional Theories of the Term Structure 323
 Expectations Theory 323
 Maturity Preference Theory 324
 Market Segmentation Theory 325
- 9.7 Determinants of Nominal Interest Rates: A Modern Perspective 325
 Problems with Traditional Theories 325
 Modern Term Structure Theory 326
 Liquidity and Default Risk 326
- 9.8 Summary and Conclusions 328
- 10. Bond Prices and Yields 337
 - 10.1 Bond Basics 338 Straight Bonds 338 Coupon Rate and Current Yield 338
 - 10.2 Straight Bond Prices and Yield to Maturity 339
 Straight Bond Prices 339
 Premium and Discount Bonds 341
 Relationships among Yield Measures 344
 A Note on Bond Price Quotes 344
 - 10.3 More on Yields 345 Calculating Yields 345 Yield to Call 346
 - 10.4 Interest Rate Risk and Malkiel's Theorems 348
 Promised Yield and Realized Yield 348
 Interest Rate Risk and Maturity 349
 Malkiel's Theorems 349
 - 10.5 Duration 352
 Macaulay Duration 352
 Modified Duration 353
 Calculating Macaulay Duration 353
 Properties of Duration 355
 - 10.6 Bond Risk Measures Based on Duration 356Dollar Value of an 01 356Yield Value of a 32nd 357
 - 10.7 Dedicated Portfolios and Reinvestment Risk 357Dedicated Portfolios 358Reinvestment Risk 359
 - 10.8 Immunization 360Price Risk versus Reinvestment Rate Risk 360Immunization by Duration Matching 361Dynamic Immunization 361
 - 10.9 Summary and Conclusions 362

PART FOUR Portfolio Management 372

- 11. Diversification and Risky Asset Allocation 372
 - 11.1 Expected Returns and Variances 373 Expected Returns 373 Calculating the Variance of Expected Returns 375
 - 11.2 Portfolios 376Portfolio Weights 376Portfolio Expected Returns 377Portfolio Variance of Expected Returns 378
 - 11.3 Diversification and Portfolio Risk 379 The Effect of Diversification: Another Lesson from Market History 379 The Principle of Diversification 382 The Fallacy of Time Diversification 382
 - 11.4 Correlation and Diversification 385
 Why Diversification Works 385
 Calculating Portfolio Risk 386
 The Importance of Asset Allocation, Part 1 388
 More on Correlation and the Risk-Return Trade-Off 390
 - 11.5 The Markowitz Efficient Frontier 392 The Importance of Asset Allocation, Part 2 392
 - 11.6 Summary and Conclusions 395
- 12. Return, Risk, and the Security Market Line 406
 - 12.1 Announcements, Surprises, and Expected Returns 407Expected and Unexpected Returns 407Announcements and News 407
 - 12.2 Risk: Systematic and Unsystematic 409 Systematic and Unsystematic Risk 409 Systematic and Unsystematic Components of Return 410
 - 12.3 Diversification, Systematic Risk, and Unsystematic Risk 411Diversification and Unsystematic Risk 411Diversification and Systematic Risk 411
 - 12.4 Systematic Risk and Beta 412 The Systematic Risk Principle 412 Measuring Systematic Risk 412 Portfolio Betas 414
 - 12.5 The Security Market Line 415 Beta and the Risk Premium 415

The Reward-to-Risk Ratio 416 The Basic Argument 416 The Fundamental Result 418 The Security Market Line 420

- 12.6 More on Beta 422
 A Closer Look at Beta 423
 Where Do Betas Come From? 424
 Another Way to Calculate Beta 426
 Why Do Betas Differ? 428
- 12.7 Extending CAPM 429A (Very) Brief History of Testing CAPM 429The Fama-French Three-Factor Model 430
- 12.8 Summary and Conclusions 432
- 13. Performance Evaluation and Risk Management 441
 - 13.1 Performance Evaluation 442
 Performance Evaluation Measures 442
 The Sharpe Ratio 443
 The Treynor Ratio 443
 Jensen's Alpha 444
 Another Method to Calculate Alpha 445
 Information Ratio 447 *R*-Squared 448
 - 13.2 Comparing Performance Measures 449Global Investment PerformanceStandards 451Sharpe-Optimal Portfolios 451
 - 13.3 Investment Risk Management 454 Value-at-Risk 454
 - 13.4 More on Computing Value-at-Risk 457
 - 13.5 Summary and Conclusions 460

PART FIVE Futures and Options 468

- 14. Futures Contracts 468
 - 14.1 Futures Contracts Basics 469Modern History of Futures Trading 469Futures Contract Features 470Futures Prices 471
 - 14.2 Why Futures? 474Speculating with Futures 474Hedging with Futures 475
 - 14.3 Futures Trading Accounts 479
 - 14.4 Cash Prices versus Futures Prices 481Cash Prices 481

Cash-Futures Arbitrage 481 Spot-Futures Parity 483 More on Spot-Futures Parity 484

- 14.5 Stock Index Futures 486
 Basics of Stock Index Futures 486
 Index Arbitrage 486
 Hedging Stock Market Risk with Futures 487
 Hedging Interest Rate Risk with
 Futures 488
 Futures Contract Delivery Options 490
- 14.6 Summary and Conclusions 490

15. Stock Options 499

- 15.1 Options on Common Stocks 500Option Basics 500Option Price Quotes 501
- 15.2 The Options Clearing Corporation 503
- 15.3 Why Options? 504
- 15.4 Stock Index Options 506Index Options: Features and Settlement 506Index Option Price Quotes 507
- 15.5 Option Intrinsic Value and "Moneyness" 509 Intrinsic Value for Call Options 509 Intrinsic Value for Put Options 510 Time Value 510 Three Lessons about Intrinsic Value 511 Show Me the Money 511
- 15.6 Option Payoffs and Profits 513
 Option Writing 513
 Option Payoffs 513
 Option Payoff Diagrams 513
 Option Profit Diagrams 515
- 15.7 Using Options to Manage Risk 517 The Protective Put Strategy 517 Credit Default Swaps 517 The Protective Put Strategy and Corporate Risk Management 518 Using Call Options in Corporate Risk Management 519
- 15.8 Option Trading Strategies 520The Covered Call Strategy 520Spreads 521Combinations 522
- 15.9 Arbitrage and Option Pricing Bounds 523The Upper Bound for Call Option Prices 523The Upper Bound for Put Option Prices 523

The Lower Bounds for Call and Put Option Prices 524 15.10 Put-Call Parity 526 Put-Call Parity with Dividends 528 What Can We Do with Put-Call Parity? 528 15.11 Summary and Conclusions 530

PART SIX Topics in Investments 540

- **16.** Option Valuation 540
 - 16.1 A Simple Model to Value Options before Expiration 541
 - 16.2 The One-Period Binomial Option Pricing Model 542

The One-Period Binomial Option Pricing Model—The Assumptions 542

The One-Period Binomial Option Pricing Model—The Setup 542

The One-Period Binomial Option Pricing Model—The Formula 543

What Is Delta? 545

16.3 The Two-Period Binomial Option Pricing Model 545

Step 1: Build a Price Tree for Stock Prices through Time 546

Step 2: Use the Intrinsic Value Formula to Calculate the Possible Option Prices at Expiration 546

Step 3: Calculate the Fractional Share Needed to Form Each Risk-Free Portfolio at the Next-to-Last Date 547

Step 4: Calculate All Possible Option Prices at the Next-to-Last Date 548

Step 5: Repeat This Process by Working Back to Today 548

- 16.4 The Binomial Option Pricing Model with Many Periods 549
- 16.5 The Black-Scholes Option Pricing Model 551
- 16.6 Varying the Option Price Input Values 554
 Varying the Underlying Stock Price 555
 Varying the Option's Strike Price 555
 Varying the Time Remaining until Option Expiration 555
 Varying the Volatility of the Stock Price 556

Varying the Interest Rate 556

16.7 Measuring the Impact of Stock Price Changes on Option Prices 557Interpreting Option Deltas 558

	16.8	Hedging Stock with Stock Options 559
		Hedging Using Call Options—The Prediction 560
		Hedging Using Call Options—The Results 560
		Hedging Using Put Options—The Prediction 560
		Hedging Using Put Options—The Results 561
	16.9	Hedging a Stock Portfolio with Stock Index Options 561
	16.10	Implied Standard Deviations 563
		CBOE Implied Volatilities for Stock Indexes 564
	16.11	Employee Stock Options 565
		ESO Features 565
		ESO Repricing 565
		ESOs at The Gap, Inc. 566
		Valuing Employee Stock Options 566
	16.12	2 Summary and Conclusions 568
17.	Proje	ecting Cash Flow and Earnings 578
	17.1	Sources of Financial Information 579
	17.2	Financial Statements 580
		The Balance Sheet 580
		The Income Statement 582
		The Cash Flow Statement 583
		Performance Ratios and Price Ratios 584
	17.3	Financial Statement Forecasting 586
		The Percentage of Sales Approach 587
		The Pro Forma Income Statement 587
		The Pro Forma Balance Sheet 588
		Scenario One 590
		Scenario Two 590
		Projected Profitability and Price Ratios 593
	17.4	Starbucks Corporation Case Study 593
		Pro Forma Income Statement 595
		Pro Forma Balance Sheet 597
		Valuing Starbucks Using Ratio Analysis 599
		Valuing Starbucks Using a Two-Stage Dividend Growth Model 601
		Valuing Starbucks: What Does the Market Say? 602
	17.5	Summary and Conclusions 603
18.	Corp	oorate and Government Bonds 615
	18.1	Corporate Bond Basics 616
	18.2	Corporate Bond Indentures 617
		Bond Seniority Provisions 618

Call Provisions 618 Put Provisions 621 Bond-to-Stock Conversion Provisions 621 Graphical Analysis of Convertible Bond Prices 623 Bond Maturity and Principal Payment Provisions 624 Sinking Fund Provisions 625 Coupon Payment Provisions 625 Protective Covenants 626 Adjustable-Rate Bonds 627 18.3 Government Bond Basics 628 18.4 U.S. Treasury Bills, Notes, Bonds, and STRIPS 628 Treasury Bond and Note Prices 631 Treasury Inflation-Protected Securities 633 18.5 U.S. Treasury Auctions 634 18.6 Federal Government Agency Securities 636 18.7 Municipal Bonds 637 Municipal Bond Features 639 Types of Municipal Bonds 640 Municipal Bond Insurance 641 Equivalent Taxable Yield 641 Taxable Municipal Bonds 643 18.8 Bond Credit Ratings 643 Why Bond Ratings Are Important 645 An Alternative to Bond Ratings 645 Junk Bonds 645 18.9 Summary and Conclusions 649 Global Economic Activity and Industry Analysis 659 19.1 Top-Down Analysis 660 19.2 Global Macroeconomic Activity 661 Real GDP 661 Business Cycles 662 Economic Indicators 664 The Global Economy and Stock Return Correlations 664 The Effects of Exchange Rates on Global Investments 665 19.3 Monitoring Jobs and the Price Level 666 Labor Market Indicators 666 The Consumer Price Index 667 19.4 Monetary and Fiscal Policy 668 Monetary Policy 668 Fiscal Policy 670

19.

- 19.5 Industry Analysis 671Identifying Sectors 671Porter's Five Forces 674
- 19.6 Summary and Conclusions 675
- 20. Mortgage-Backed Securities (Web site only) 20-1
 - 20.1 A Brief History of Mortgage-Backed Securities 20-2
 - 20.2 Fixed-Rate Mortgages 20-2 Fixed-Rate Mortgage Amortization 20-3 Fixed-Rate Mortgage Prepayment and Refinancing 20-5
 - 20.3 Government National Mortgage Association 20-10 GNMA Clones 20-10
 - 20.4 Public Securities Association Mortgage Prepayment Model 20-11
 - 20.5 Cash Flow Analysis of GNMA Fully Modified Mortgage Pools 20-13

Macaulay Durations for GNMA Mortgage-Backed Bonds 20-14 20.6 Collateralized Mortgage Obligations 20-16 Interest-Only and Principal-Only Mortgage Strips 20-17 Sequential Collateralized Mortgage Obligations 20-19

Protected Amortization Class Bonds 20-21

- 20.7 Yields for Mortgage-Backed Securities and Collateralized Mortgage Obligations 20-23
- 20.8 Summary and Conclusions 20-24

APPENDIXES

- A Answers to Test Your Investment Quotient Questions 682
- Answers to Selected Questions and Problems 686
- C Key Equations 688

Index 692

This page intentionally left blank

chapter 1 A Brief History of Risk and Return

"All I ask is for the chance to prove that money can't make me happy."

–Spike Milligan

Learning Objectives

To become a wise investor (maybe even one with too much money), you need to know:

- How to calculate the return on an investment using different methods.
- 2. The historical returns on various important types of investments.
- 3. The historical risks on various important types of investments.
- 4. The relationship between risk and return.

Who wants to be a millionaire? Actually, anyone can retire as a millionaire. How? Consider this: Suppose you, on your 25th birthday, invest \$3,000. You have the discipline to invest \$3,000 on each of your next 39 birthdays until you retire on your 65th birthday. How much will you have? The answer might surprise you. If you earn 10 percent per year, you will have about \$1.46 million. Are these numbers realistic? Based on the history of financial markets, the answer appears to be yes. For example, over the last 87 or so years, the widely followed Standard & Poor's Index of large-company common stocks has actually yielded about 12 percent per year.

The study of investments could begin in many places. After thinking it over, we decided that a brief history lesson is in order, so we start our discussion of risk and return by looking back at what has happened to investors in U.S. financial markets since 1925. In 1931, for example, the stock market lost 43 percent of its value. Just two years later, the market reversed itself and gained 54 percent. In more recent times, the stock market lost about 25 percent of

CFA[™] Exam Topics in This Chapter:

- 1 Discounted cash flow applications (L1, S2)
- 2 Statistical concepts and market returns (L1, S2)
- 3 Common probability distributions (L1, S3)
- 4 Sampling and estimation (L1, S3)
- 5 Dividend and share repurchases: Analysis (L2, S8)
- **6** Evaluating portfolio performance (L3, S17)

Go to www.mhhe.com/jmd7e for a guide that aligns your textbook with CFA readings.

its value on October 19, 1987, alone, and it gained almost 40 percent in 1995. From 2003 through 2007, the market gained about 80 percent. In 2008, the market fell almost 40 percent. In 2009, the market reversed course again, returning almost 20 percent.

So what should you, as a stock market investor, expect when you invest your own money? In this chapter, we study more than eight decades of market history to find out.

In this chapter, we present the historical relation between risk and return. As you will see, this chapter has a lot of very practical information for anyone thinking of investing in financial assets such as stocks and bonds. For example, suppose you were to start investing in stocks today. Do you think your money would grow at an average rate of 5 percent per year? Or 10 percent? Or 20 percent? This chapter gives you an idea of what to expect (the answer may surprise you). The chapter also shows how risky certain investments can be, and it gives you the tools to think about risk in an objective way.

Our primary goal in this chapter is to see what financial market history can tell us about risk and return. Specifically, we want to give you a perspective on the numbers. What is a high return? What is a low return? More generally, what returns should we expect from financial assets such as stocks and bonds, and what are the risks from such investments? Beyond this, we hope that by studying what *did* happen in the past, we will at least gain some insight into what *can* happen in the future.

The history of risk and return is made day by day in global financial markets. The Internet is an excellent source of information on financial markets. Visit our Web site (at www.mhhe .com/jmd7e) for suggestions on where to find information on recent financial market events. We will suggest other sites later in the chapter.

Not everyone agrees on the value of studying history. On the one hand, there is philosopher George Santayana's famous comment, "Those who do not remember the past are condemned to repeat it." On the other hand, there is industrialist Henry Ford's equally famous comment, "History is more or less bunk." These extremes aside, perhaps everyone would agree with Mark Twain, who observed, with remarkable foresight (and poor grammar), that "October. This is one of the peculiarly dangerous months to speculate in stocks in. The others are July, January, September, April, November, May, March, June, December, August, and February."

Two key observations emerge from a study of financial market history. First, there is a reward for bearing risk, and, at least on average, that reward has been substantial. That's the good news. The bad news is that greater rewards are accompanied by greater risks. The fact that risk and return go together is probably the single most important fact to understand about investments, and it is a point to which we will return many times.

1.1 Returns

We wish to discuss historical returns on different types of financial assets. First, we need to know how to compute the return from an investment. We will consider buying shares of stock in this section, but the basic calculations are the same for any investment.

DOLLAR RETURNS

If you buy an asset of any type, your gain (or loss) from that investment is called the *return* on your investment. This return will usually have two components. First, you may receive some cash directly while you own the investment. Second, the value of the asset you purchase may change. In this case, you have a capital gain or capital loss on your investment.¹

To illustrate, suppose you purchased 200 shares of stock in Harley-Davidson (ticker symbol: HOG) on January 1. At that time, Harley was selling for \$50 per share, so your 200 shares cost you \$10,000. At the end of the year, you want to see how you did with your investment.

¹ As a practical matter, what is and what is not a capital gain (or loss) is determined by the Internal Revenue Service. Even so, as is commonly done, we use these terms to refer to a change in value.

The first thing to consider is that over the year, a company may pay cash dividends to its shareholders. As a stockholder in Harley, you are a part owner of the company, and you are entitled to a portion of any money distributed. So if Harley chooses to pay a dividend, you will receive some cash for every share you own.

In addition to the dividend, the other part of your return is the capital gain or loss on the stock. This part arises from changes in the value of your investment. For example, consider these two cases:

	Case 1	Case 2
Ending Stock Price	\$ 55.60	\$ 39.80
January 1 value	\$10,000	\$10,000
December 31 value	\$11,120	\$ 7,960
Dividend income	\$ 80	\$ 80
Capital gain or loss	\$ 1,120	-\$ 2,040

At the beginning of the year, on January 1, the stock was selling for \$50 per share. As we calculated above, your total outlay for 200 shares is \$10,000. Over the year, Harley paid dividends of \$.40 per share. By the end of the year, then, you received dividend income of

Dividend income = $40 \times 200 = 80$

In Case 1, suppose that as of December 31, a HOG share was selling for \$55.60, meaning that the value of your stock increased by \$5.60 per share. Your 200 shares would be worth \$11,120, so you have a capital gain of

Capital gain =
$$(\$55.60 - \$50) \times 200 = \$1,120$$

On the other hand, if the price had dropped to, say, \$39.80 (Case 2), you would have a capital loss of

Capital loss = $(\$39.80 - \$50) \times 200 = -\$2,040$

Notice that a capital loss is the same thing as a negative capital gain.

The **total dollar return** on your investment is the sum of the dividend income and the capital gain (or loss):

Total dollar return = Dividend income + Capital gain (or loss)

In Case 1, the total dollar return is thus given by

Total dollar return = 80 + 1,120 = 1,200

Overall, between the dividends you received and the increase in the price of the stock, the value of your investment increased from 10,000 to 10,000 + 1,200 = 11,200.

A common misconception often arises in this context. Suppose you hold on to your Harley-Davidson stock and don't sell it at the end of the year. Should you still consider the capital gain as part of your return? Isn't this only a "paper" gain and not really a cash gain if you don't sell it?

The answer to the first question is a strong yes, and the answer to the second is an equally strong no. The capital gain is every bit as much a part of your return as the dividend, and you should certainly count it as part of your return. The fact that you decide to keep the stock and don't sell (you don't "realize" the gain) is irrelevant because you could have converted it to cash if you had wanted to. Whether you choose to do so is up to you.

total dollar return

The return on an investment measured in dollars that accounts for all cash flows and capital gains or losses.

Our favorite investments Web site is Yahoo! Finance at finance.yahoo.com Visit this site and look around! After all, if you insist on converting your gain to cash, you could always sell the stock and immediately reinvest by buying the stock back. There is no difference between doing this and just not selling (assuming, of course, that there are no transaction costs or tax consequences from selling the stock). Again, the point is that whether you actually cash out and buy pizzas (or whatever) or continue to hold the investment doesn't affect the return you actually earn.

PERCENTAGE RETURNS

It is usually more convenient to summarize information about returns in percentage terms than in dollar terms, because that way your return doesn't depend on how much you actually invested. With percentage returns the question we want to answer is: How much do we get *for each dollar* we invest?

To answer this question, let P_t be the price of the stock at the beginning of the year. Let D_{t+1} be the dividend paid on the stock during the year. The following cash flows are the same as those shown earlier, except that we have now expressed everything on a per-share basis:

	Case 1	Case 2
January 1 stock price, P _t	\$50.00	\$50.00
December 31 stock price, P_{t+1}	\$55.60	\$39.80
Dividend income, D _{t + 1}	\$.40	\$.40
Capital gain or loss	\$ 5.60	-\$10.20

In our example, the price at the beginning of the year was \$50 per share and the dividend paid during the year on each share was \$.40. If we divide the dividend by the beginning stock price, the result is the **dividend yield**:

Dividend yield =
$$D_{t+1} / P_t$$

= \$.40 / \$50 = .0080 = 0.80% (1.1)

This calculation says that for each dollar we invested we received 0.80 cents in dividends.

The second component of our percentage return is the **capital gains yield**. This yield is calculated as the change in the price during the year (the capital gain) divided by the beginning price. With the Case 1 ending price, we get:

Capital gains yield =
$$(P_{t+1} - P_t) / P_t$$

= (\$55.60 - \$50.00) / \$50.00
= \$5.60 / \$50 = .1120 = 11.20% (1.2)

(1.3)

This 11.20 percent yield means that for each dollar invested we got about 11 cents in capital gains (HOG heaven).

Putting it all together, per dollar invested, we get 0.80 cents in dividends and 11.20 cents in capital gains for a total of 12.00 cents. Our **total percent return** is 12 cents on the dollar, or 12.00 percent. When a return is expressed on a percentage basis, we often refer to it as the *rate of return*, or just "return," on the investment. Notice that if we combine the formulas for the dividend yield and capital gains yield, we get a single formula for the total percentage return:

Percentage return = Dividend yield + Capital gains yield = $D_{t+1} / P_t + (P_{t+1} - P_t) / P_t$ = $(D_{t+1} + P_{t+1} - P_t) / P_t$

To check our calculations, notice that we invested \$10,000 and ended up with \$11,200. By what percentage did our \$10,000 increase? As we saw, our gain was 11,200 - 10,000 = 1,200. This is an increase of \$1,200 / \$10,000, or 12.00 percent.

dividend yield

The annual stock dividend as a percentage of the initial stock price.

capital gains yield

The change in stock price as a percentage of the initial stock price.

total percent return

The return on an investment measured as a percentage that accounts for all cash flows and capital gains or losses.

WORK THE WEB

To look up information on common stocks using the Web, you need to know the "ticker" symbol for the stocks in which you are interested. You can look up ticker symbols in many places, including one of our favorite sites, finance. yahoo.com. Here we have looked up (using the "Symbol Lookup" link) and entered ticker symbols for some well-known "tech" stocks: Dell, Cisco, Intel, and Microsoft.

YAHOO! FINANCE	Q Search
dell csco into msft GET QUOTES Finance Search	Mon, Feb 1, 2010, 10:15AM ET

Once we hit "Get Quotes," this is what we got:

SYMBOL	TIME & PRICE		CHG &	% CHG	DAY'S LO	N & HIGH	VOLUME	AVG VOL	MKT CAP
DELL	04:00pm EST	13.81	0.10	+0.73%	13.68	13.83	20,541,699	38,353,100	23.99B
csco	04:00pm EST	20.99	0.00	+0.00%	20.92	21.07	44,439,782	38,169,000	111.62B
INTC	04:00pm EST	21.115	-0.11	-0.54%	20.99	21.30	33,809,049	48,423,300	104.90B
MSFT	04:00pm EST	28.01	-0.03	-0.11%	27.88	28.16	49,651,288	51,124,300	234.62B

As you can see, we get the price for each stock, along with information about the change in price and volume (number of shares traded). You will find a lot of links to hit and learn more, so have at it!

EXAMPLE 1.1

Calculating Percentage Returns

Suppose you buy some stock in Concannon Plastics for \$35 per share. After one year, the price is \$49 per share. During the year, you received a \$1.40 dividend per share. What is the dividend yield? The capital gains yield? The percentage return? If your total investment was \$1,400, how much do you have at the end of the year?

Your \$1.40 dividend per share works out to a dividend yield of

Dividend yield =
$$D_{t+1} / P_t$$

= \$1.40 / \$35
= 4%

The per-share capital gain is \$14, so the capital gains yield is

Capital gains yield = $(P_{t+1} - P_t) / P_t$ = (\$49 - \$35) / \$35 = \$14 / \$35 = 40%

The total percentage return is thus 4% + 40% = 44%.

If you had invested \$1,400, you would have \$2,016 at the end of the year. To check this, note that your \$1,400 would have bought you \$1,400 / \$35 = 40 shares. Your 40 shares would then have paid you a total of $40 \times $1.40 = 56 in cash dividends. Your \$14 per share gain would give you a total capital gain of $$14 \times 40 = 560 . Add these together and you get \$616, which is a 44 percent total return on your \$1,400 investment.

A NOTE ON ANNUALIZING RETURNS

So far, we have only considered annual returns. Of course, the actual length of time you own an investment will almost never be exactly a year. To compare investments, however, we will usually need to express returns on a per-year or "annualized" basis, so we need to do a little bit more work.

For example, suppose you bought 200 shares of Lowe's Companies, Inc. (LOW), at a price of \$30 per share. In three months, you sell your stock for \$31.50. You didn't receive any dividends. What is your return for the three months? What is your annualized return?

In this case, we say that your *holding period*, which is the length of time you own the stock, is three months. With a zero dividend, you know that the percentage return can be calculated as:

Percentage return = $(P_{t+1} - P_t)/P_t = (\$31.50 - \$30)/\$30 = .0500 = 5.00\%$

This 5.00 percent is your return for the three-month holding period, but what does this return amount to on a per-year basis? To find out, we need to convert this to an annualized return, meaning a return expressed on a per-year basis. Such a return is often called an **effective annual return**, or **EAR** for short. The general formula is this:

 $1 + EAR = (1 + holding period percentage return)^m$ (1.4)

where *m* is the number of holding periods in a year.

In our example, the holding period percentage return is 5.00 percent, or .0500. The holding period is three months, so there are four (12 months/3 months) periods in a year. We calculate the annualized return, or *EAR*, as follows:

> $1 + EAR = (1 + \text{holding period percentage return})^m$ = (1 + .0500)⁴ = 1.2155

So, your annualized return is 21.55 percent.

EXAMPLE 1.2 A "QWEST" for Returns

Suppose you buy some stock in Qwest (no, that's not a typo, that's how the company spells it) at a price of \$8 per share. Four months later, you sell for \$8.40 per share. No dividend is paid. What is your annualized return on this investment? For the four-month holding period, your return is:

Percentage return = $(P_{t+1} - P_t) / P_t = (\$8.40 - \$8) / \$8 = .05 = 5\%$

There are three four-month periods in a year, so the annualized return is:

 $1 + EAR = (1 + holding period percentage return)^m = (1 + .05)^3 = 1.1576$

Subtracting the one, we get an annualized return of .1576, or 15.76 percent.

EXAMPLE 1.3 More Annualized Returns

Suppose you buy some stock in Johnson & Johnson (JNJ) at a price of \$60 per share. Three *years* later, you sell it for \$64.50. No dividends were paid. What is your annualized return on this investment?

The situation here is a bit different because your holding period is now longer than a year, but the calculation is basically the same. For the three-year holding period, your return is:

Percentage return = $(P_{t+1} - P_t) / P_t = (\$64.50 - \$60) / \$60 = .075 = 7.5\%$

(continued)

effective annual return (EAR)

The return on an investment expressed on a per-year, or "annualized," basis. How many three-year holding periods are there in a single year? The answer is one-third, so m in this case is 1/3. The annualized return is:

 $1 + EAR = (1 + holding period percentage return)^m$ $= (1 + .075)^{1/3}$ = 1.0244

Subtracting the one, we get an annualized return of .0244, or 2.44 percent.

Now that you know how to calculate returns on a hypothetical stock, you should calculate returns for real stocks. The nearby *Work the Web* box using finance.yahoo.com describes how to begin. Meanwhile, in the next several sections, we will take a look at the returns that some common types of investments have earned over the last 87 years.

\checkmark			
	CHECK THIS	1.1a 1.1b 1.1c	What are the two parts of total return? What is the difference between a dollar return and a percentage return? Why are percentage returns usually more convenient? What is an effective annual return (EAR)?

1.2 The Historical Record

We now examine year-to-year historical rates of return on five important categories of financial investments. These returns can be interpreted as what you would have earned if you had invested in portfolios of the following asset categories:

- 1. Large-company stocks. The large-company stock portfolio is based on the Standard & Poor's (S&P's) 500 Index, which contains 500 of the largest companies (in terms of total market value of outstanding stock) in the United States.
- **2.** Small-company stocks. This is a portfolio composed of stock of smaller companies, where "small" corresponds to the smallest 20 percent of the companies listed on the New York Stock Exchange, again as measured by market value of outstanding stock.
- **3.** Long-term corporate bonds. This is a portfolio of high-quality bonds with 20 years to maturity.
- **4.** Long-term U.S. government bonds. This is a portfolio of U.S. government bonds with 20 years to maturity.
- **5.** U.S. Treasury bills. This is a portfolio of Treasury bills (T-bills for short) with a three-month maturity.

If you are not entirely certain what these investments are, don't be overly concerned. We will have much more to say about each in later chapters. For now, just accept that these are some important investment categories. In addition to the year-to-year returns on these financial instruments, the year-to-year percentage changes in the Consumer Price Index (CPI) are also computed. The CPI is a standard measure of consumer goods price inflation. We discuss the CPI in more detail in a later chapter.

Here is a bit of market jargon for you. A company's *total market capitalization* (or market "cap" for short) is equal to its stock price multiplied by the number of shares of stock. In other words, it's the total value of the company's stock. Large companies are often called "large-cap" stocks, and small companies are called "small-cap" stocks. We'll use these terms frequently.



Annual historical financial market data can be downloaded (but not for free) at www.globalfinancialdata.com



Source: Global Financial Data (www.globalfinancialdata.com) and Professor Kenneth R. French, Dartmouth College.

A FIRST LOOK

Before examining the different portfolio returns, we first take a look at the "big picture." Figure 1.1 shows what happened to \$1 invested in these different portfolios at the beginning of 1926 and held over the 87-year period ending in 2012 (for clarity, the long-term corporate bonds are omitted). To fit all the information on a single graph, some modification in scaling is used. As is commonly done with financial time series, the vertical axis is scaled so that equal distances measure equal percentage (as opposed to dollar) changes in value. Thus, the distance between \$10 and \$100 is the same as that between \$100 and \$1,000, since both distances represent the same 900 percent increases.

Looking at Figure 1.1, we see that the small-company investment did the best overall. Every dollar invested grew to a remarkable \$21,997.36 over the 87 years. The larger common stock portfolio did less well; a dollar invested in it grew to \$3,247.50.

At the other end, the T-bill portfolio grew to only \$22.39. This is even less impressive when we consider the inflation over this period. As illustrated, the increase in the price level was such that \$12.83 is needed just to replace the original \$1.

Given the historical record, why would anybody buy anything other than small-company stocks? If you look closely at Figure 1.1, you will probably see the answer—risk. The T-bill portfolio and the long-term government bond portfolio grew more slowly than did the stock portfolios, but they also grew much more steadily. The small stocks ended up on top, but, as you can see, they grew quite erratically at times. For example, the small stocks were the



FIGURE 1.2



Sources: Jeremy J. Siegel, Stocks for the Long Run, 3rd ed. (New York: McGraw-Hill, 2003); update through 2009 provided by Jeremy J. Siegel; update through 2012 from Global Financial Data (www.globalfinancialdata.com); and Professor Kenneth R. French, Dartmouth College.

worst performers for about the first 10 years and had a smaller return than long-term government bonds for almost 15 years.

A LONGER RANGE LOOK

The data available on the stock returns before 1925 are not comprehensive, but it is nonetheless possible to trace reasonably accurate returns in U.S. financial markets as far back as 1801. Figure 1.2 shows the values, in 2012, of \$1 invested since 1801 in stocks, long-term bonds, short-term bills, and gold. The CPI is also included for reference.

Inspecting Figure 1.2, we see that \$1 invested in stocks grew to an astounding \$14.6 million over this 212-year period. During this time, the returns from investing in stocks dwarf those earned on other investments. Notice also in Figure 1.2 that, after 170 years, gold has managed to outpace inflation beginning in the 1970s.

What we see thus far is that there has been a powerful financial incentive for long-term investing. The real moral of the story is this: Get an early start!

A CLOSER LOOK

To illustrate the variability of the different investments and inflation, Figures 1.3 through 1.6 plot the year-to-year percentage returns in the form of vertical bars drawn from the horizontal axis. The height of a bar tells us the return for the particular year. For example, looking at the long-term government bonds (Figure 1.5), we see that the largest historical return (47.14 percent) occurred in 1982. This year was a good year for bonds. In comparing these charts, notice the differences in the vertical axis scales. With these differences in mind,